## Selection of P3 delivery methods for Sustainable Social Infrastructure Projects Using the Analytical Hierarchical Process

Nita Semgalawe

# A Thesis

# In the Department of

# Building, Civil and Environmental Engineering

Presented in Partial Fulfillment of the Requirements for the Degree of Master of Applied Science (Building Engineering)

> at Concordia University Montréal, Québec, Canada

> > April 2024

© Nita Semgalawe, 2024

## **CONCORDIA UNIVERSITY**

## SCHOOL OF GRADUATE STUDIES

This is to certify that the thesis prepared

- By: Nita Semgalawe
- Entitled: Selection of P3 delivery methods for Sustainable Social Infrastructure Projects Using the Analytical Hierarchical Process

and submitted in partial fulfillment of the requirements for the degree of

## Master of Applied Science (Building Engineering)

Complies with the regulations of the university and meets the accepted standards with respect to originality and quality.

Signed by the final examining committee:

		Chair
	Dr. Sang Hyeok Han	
		External Examiner
	Dr. Govind Gopakumar	
		Examiner
	Dr. Sang Hyeok Han	
		Thesis Supervisor
	Dr. Osama Moselhi	
Approv	red by:	
	Dr. Chunjiang An	Graduate Program Director
April 16	5 <sup>th</sup> 2024	
	Dr. Mourad Debbabi	Dean of Gina Cody School of Engineering and Computer Science

## Abstract

Selection of P3 delivery methods for Sustainable Social Infrastructure Using the Analytical Hierarchical Process

#### Nita Semgalawe

This thesis studies the necessary shift in screening practices of public-private partnerships (P3) projects in Canada, moving beyond traditional qualitative criteria to include broader environmental, social, and governance (ESG) project objectives. The current P3 screening, while effective, needs adaptation to align with Canadian societal and environmental infrastructure goals. In response, this thesis focuses on three objectives aimed at improving social infrastructure P3 procurement and promoting sustainable and responsible project management practices for these projects. Firstly, it identifies and describes Canadian-specific ESG criteria important for ensuring responsible sustainability in delivering social infrastructure projects. Secondly, it develops an ESG-PPP screening matrix to evaluate social infrastructure projects based on responsible sustainability thresholds, determining their suitability for P3 procurement. Thirdly, it implements a multi-criteria analysis using the Analytic Hierarchy Process (AHP) to determine the most appropriate P3 model for social infrastructure projects, considering the identified ESG criteria and quantitative value-formoney criterion. The AHP-PPP selection tool is applied to three case studies analyzing two AHP scales to assess their consistency ratios and the reliability of the P3 selection results. The results indicate that the balanced-n scale exhibit lower inconsistency ratios compared to the fundamental AHP scale, and decisions on P3 options remained consistent across all case studies using both scales, suggesting that the Fundamental AHP scale remains reliable if decision-makers accurately reflect the relative importance of P3 options. Overall, this thesis addresses the increasing need for sustainable and responsible management of social infrastructure projects in Canada by integrating ESG factors into the current procurement process.

iii

# Acknowledgements

I am sincerely grateful to my supervisor, **Dr. Osama Moselhi**, for providing me with valuable guidance and unwavering support throughout the course of my research. His guidance has played an important role in shaping the depth and quality of this research work.

Also, I would also like to thank my family and friends for their constant encouragement and support.

# Dedication

То

Juana, Dr. Zainab Semgalawe, and Dr. Kristina Lugangira, your support and encouragement have given me strength, purpose, and motivation.

# TABLE OF CONTENTS

LIST	OF FIGURES	viii
LIST	OF TABLES	ix
LIST	OF ABBREVIATIONS	x
Chapter	1: Introduction	1
1.1.	Overview	1
1.2.	Problem Statement	4
1.3.	Scope and Objectives	5
1.4.	Thesis Organization	6
Chapter	2: Literature Review	8
2.1.	Introduction	8
2.2.	Objectives of social infrastructure projects in Canada	8
2.3.	Public-private partnership models in Canada	17
2	2.3.1. Procurement of a P3 model for social infrastructure in Canada	24
2	2.3.2. PPP project governance in Canada	28
2.4.	Sustainable and Responsible P3 project management	36
2.5.	Summary of identified gaps in the literature	50
Chapter	3: Methodology	53
3.1.	Introduction	53
3.2.	Development of the Screening Matrix	53
3.3.	Analytical Hierarchy Process for procurement of P3 options	60

Chapter 4: Application of AHP-PPP Excel Tool	74
4.1. Introduction	74
4.2. Case Study Selection	74
4.2.1. Case study 1: South Niagara Hospital	74
4.2.2. Case study 2: Finch West Light Rail Transit	80
4.2.3. Case study 3: Saskatchewan Joint Use School Projects	84
4.3. Analysis and Discussion	87
Chapter 5: Conclusion and Recommendations	90
5.1. Conclusion	90
5.2. Recommendations	92
Appendix I	93
Appendix II	94
Appendix III	96
Appendix IV1	01
References1	07

# LIST OF FIGURES

Figure 1. The number of completed social infrastructure projects in Canada by 2023 [9]2
Figure 2. The number of pipeline social infrastructure projects in Canada from 2023 [9]2
Figure 3. Partnership Structure in a Typical P3 Model in Canada18
Figure 4. Traditional PDM and P3 models in Canada, adapted from [57]18
Figure 5. Ontario's P3 models across Different Social Infrastructure Projects20
Figure 6. Alberta's P3 models across Different Social Infrastructure Projects
Figure 7. Saskatchewan's P3 models across Different Social Infrastructure Projects21
Figure 8. P3 models across different Social Infrastructure Projects: Québec, British Columbia,
New Brunswick, and Manitoba [9]22
Figure 9. P3 models across Different Social Infrastructure Projects: NL, Nova Scotia, PEI, and
Yukon [9]23
Figure 10. Procurement Phases of a Social Infrastructure P3 Model in Canada24
Figure 11. The Two-Stage Procurement Process of a P3 Model in Canada25
Figure 12. Existing evaluation framework for assessing P3 model potential in Canada [68]25
Figure 13 Existing P3 Decision Flow Chart for Social Infrastructure Projects in Canada27
Figure 14. Stakeholder Structure in P3s in Canada [57]
Figure 15. Community Benefit Agreements in Canadian Social Infrastructure Projects
Figure 16. Environmental Certification in Social Infrastructure Projects in Canada
Figure 17. Third-Party Certification in Canadian Infrastructure Projects
Figure 18. Decision Flow Chart for use of ESG Screening Matrix in P3 Model Procurement46
Figure 19. Literature review gaps
Figure 20. Need for Integrating ESG Factors in Procurement evaluation of a P3 model51
Figure 21. Process Flow Diagram for Screening Matrix Development
Figure 22. ESG dimension and evaluative criteria for the ESG-PPP screening matrix56
Figure 23. AHP Development Process Flow Diagram61
Figure 24 Hierarchical Structure for Selecting Sustainable and Responsible P3 Models65
Figure 25 Key Stages Following Hierarchy Structure Establishment
Figure 26. Different types of AHP scales67
Figure 27. Pairwise matrix
Figure 28 Normalised Matrix70
Figure 29. South Niagara Hospital: Priority weights of P3 models and each criterion79
Figure 30. Finch West Light Rail Transit: Priority weights of P3 models and each criterion83
Figure 31. Saskatchewan Joint Use School Projects: Priority weights of P3 models and each
criterion

## LIST OF TABLES

Table 1. Existing P3 suitability criterion and sub-criteria for social infrastructure projects	3
Table 2. Key objectives for Social Housing in Canada.	. 10
Table 3. Key objectives for Schools in Canada	. 11
Table 4. Key objectives for Hospitals in Canada	.13
Table 5. Key objectives for public transit in Canada	. 15
Table 6. Key objectives for recreation and culture in Canada	. 16
Table 7. Category of PDM used in Canada for social infrastructures	. 17
Table 8. Types of P3 models in Canada	.19
Table 9. Common Factors and Key Thresholds for Considering P3s in Canadian Provinces	.26
Table 10. Government Mechanisms and Considerations in Canadian PPPs	.29
Table 11. Findings and Recommendations for Sustainable and Responsible PM	. 37
Table 12 Qualitative ESG Criteria for Canadian Social Infrastructure Projects	.47
Table 13. Criteria Weighting Methods	. 57
Table 14. Criteria Weighting and Scoring	. 58
Table 15. P3 Procurement Decision Thresholds for Responsible Sustainability	. 59
Table 16. The comparative scales	.68
Table 17. Consistency indices for a randomly generated matrix [134]	.71
Table 18. Key Objectives and Proposed ESG Evaluation Criteria for the Hospital Project	.76
Table 19. Verbal Preference to Numerical Conversion Input (c) used in AHP-PPP Excel Tool	.77
Table 20. South Niagara Hospital: Local criteria priorities	.77
Table 21. South Niagara Hospital: Local alternative priorities with respect to each criterion	.78
Table 22. South Niagara Hospital: Overall priority weights of P3 model option	.79
Table 23. Key Objectives and Proposed ESG Evaluation Criteria for the Transit Project	.81
Table 24. Finch West Light Rail Transit: Local Criteria Priorities	.81
Table 25. Finch West Light Rail Transit: Local Alternative Priority with respect to Each Criterior	n
	.82
Table 26. Finch West Light Rail Transit: Overall Priority weights of P3 model option	.83
Table 27. Key objectives and ESG consideration for Saskatchewan Joint Use School Projects	.85
Table 28. Saskatchewan Joint Use School Projects: Local criteria priorities	.85
Table 29. Saskatchewan Joint Use School Projects: Local alternative priorities with respect to	
each criterion	.86
Table 30. Saskatchewan Joint Use School Projects: Overall priority weights of P3 model option	n
	.87

# LIST OF ABBREVIATIONS

The Table below lists abbreviations and acronyms frequently used in this thesis. It does not include nonstandard acronyms used occasionally to abbreviate mathematical variables.

PDM	Project delivery model
PM	Project management
PPP/P3	Public-private partnership
ESG	Environmental, Social and Governance
VfM	Value for Money
MCA	Multi Criteria Analysis
AHP	Analytical Hierarchy Process
DB	Design-build
DBB	Design-bid-build
DBF	Design-build-finance
DBFM	Design-build-finance-maintain
DBFOM	Design-build-finance-operate-maintain
PR. DBFM	Progressive Design-build-finance-maintain
PR. DBFOM	Progressive Design-build-finance-operate-maintain
PPP-ESG	Public private partnership - Environmental, Social and Governance
AHP- PPP	Analytical Hierarchy Process- Public-private partnership
CCPPP	Canadian Council for Public private partnership
CBA	Community Benefits Agreement
LEED	Leadership in Energy and Environmental Design
CCDC	Canadian Construction Documents Committee
CAGBC	Canadian Green Building Council
TRUE	Total Resource Use and Efficiency
ZCB	Zero Carbon Building
NL	Newfoundland and Labrador
PEI	Prince Edward Island
VBA	Visual basics for Applications
PWGSC	Public Works and Government Service Canada
SMEs	Small and Medium Sized Enterprises

# Chapter 1: Introduction

#### 1.1. Overview

Growing concerns about climate change, social inequalities, and limited financial resources emphasize the urgent need for sustainable development in the construction industry. Canada's response involves implementing the Canada Investing Plan, directing funds to projects that foster inclusive communities, generate job opportunities, and develop robust infrastructure systems [1]. In 2021, over \$71 billion has been allocated for key infrastructure areas like public transit, green projects, community facilities, rural development, and COVID-19 resilience [1] [2] [3]. This substantial financial commitment demonstrates that Canada is committed to prioritizing social and environmental aspects in infrastructure projects, with a primary focus on sustainable development.

Sustainable development, as defined by the Canadian Federal Sustainable Development Act, is the simultaneous fulfillment of present needs while preserving the capacity of future generations to meet their own needs [4] similar to the definition provided by the Brundtland Commission Report [5]. Thus, adopting a sustainable development approach when procuring infrastructure projects in Canada is important to ensure that projects are not only delivered within set budgets and timelines, but also contribute to the long-term preservation of the environmental, social, and economic well-being of future generations.

Social infrastructure projects constitute the largest proportion of the total infrastructure projects undertaken in Canada. They comprise physical facilities and spaces where can come together, engage in social activities, and access important services like education and healthcare. These spaces support human interaction, community engagement, and social development, making communities more livable and meaningful. They play an important role in meeting people's social needs, promoting connections, and enhancing overall well-being [6] [7] [8]. As seen in

1

Figures 1 and 2 below, they include hospitals, schools, public transit, housing, recreation, and culture buildings.



Figure 1. The number of completed social infrastructure projects in Canada by 2023 [9].





The procurement process for social infrastructure projects follows a two-stage process. Initially, a qualitative assessment screens various project delivery models (PDM) against specific objectives established from the project's description and needs assessment. These objectives, in conjunction with project constraints, serve as the evaluative criteria for the project delivery models. Table 1 below displays the existing most commonly used evaluating criteria for social infrastructures in Canada.

The second stage, known as the quantitative assessment, involves conducting a detailed VfM analysis on a shortlist of options that meet the qualitative project requirements. Both phases aim to compare the benefits and risks associated with potential project delivery models, evaluating their suitability for achieving the project objectives. However, the primary selection of the project delivery model takes place during the quantitative stage. This stage emphasizes quantitative factors, especially financial considerations.

Existing Qualitative Criteria	Existing Qualitative Sub-Criteria	Existing Quantitative Criteria
Project characteristics and scope	<ol> <li>Project size: 25-35 years,</li> <li>\$100 million</li> </ol>	Base costs:
		Construction cost
	2. Project complexity and innovation need	Operating cost
	3. Project phases and	Maintenance cost
	integration (O&M)	Lifecycle cost
	4. Project Type: New Construction	Competitive neutrality
Cost and historical factors	<ol> <li>Project financing and funding need</li> </ol>	Financing cost
	<ol> <li>Past project cost comparison and VfM achieved</li> </ol>	
Risk factors	1. Risk allocation	Retained risk cost value
	2. Risk transfer	Premium cost
Market capacity factors	<ol> <li>Private sector interest Possibility of a competitive procurement process</li> </ol>	
	2. Public sector capacity	

Table 1. Existing P3 suitability criterion and sub-criteria for social infrastructure projects.

Existing Qualitative Criteria	Existing Qualitative Sub-Criteria	Existing Quantitative Criteria
Contract factors	<ol> <li>Quantifiable contract performance output</li> </ol>	Ancillary cost
Stakeholder engagement (Limited)	1. Collaboration with Stakeholders	
	2. Acceptability	

The most common project delivery models compared during the procurement process are traditional ones like Design-Bid-Build (DBB) and Design-Build (DB), along with construction period-only partnerships with private finance like Design-Build-and-Finance (DBF), versus long-term partnerships such as Public-Private Partnership (P3) models. Social infrastructure projects often use PPP (also referred to as P3) models like Design-Build-Finance-and-Maintain (DBFM), Design-Build-Finance-Maintain-and-Operate (DBFOM), and the P3 bundle model. Additionally, the progressive P3 model is emerging as a new procurement approach [10] [11] for social infrastructure projects, particularly for transit projects in Ontario.

## 1.2. Problem Statement

The procurement process of P3 model for the delivery of social infrastructure projects lacks a comprehensive approach aligned with Canada's commitment to sustainable development. There is a need to have a more relevant procurement process that is evolving with Canada's objectives of sustainable development in the infrastructure sector and addressing transparency issues [11] impacting the P3 contract management process. The success of a P3 project goes beyond its financial considerations, challenging the traditional notion [12] [13]. Recent research on sustainable project management [14] suggests that achieving non-financial outcomes related to social and environmental issues is equally important for the overall success of construction projects during their management and delivery. Neglecting these social and environmental considerations in evaluating P3 model limits the chance to improve community benefits relevant to different community members and groups including indigenous peoples, thereby increasing the likelihood of social disparities and community dissatisfaction [2] [15]. Additionally, issues with governance involving transparency, and ethics in managing social infrastructure projects have led to cost overruns and cases of corruption and fraud [16], together with the perceived lack of accountability as it operates under a consortium of private companies and the use of tight-schedule procurement process timeline [17] limit the community involvement and stakeholder engagement [18]. These examples highlight the need for adapting the existing procurement process for infrastructure projects specifically social infrastructure, which directly serves and engages the community. This adaptation is essential to ensure a more ethical and inclusive approach to delivering P3 projects in Canada.

Therefore, integrating environmental, social and governance factors into P3 assessments could lead to improved outcomes and value creation [19], a potential that has largely been overlooked. A procurement decision process that integrates qualitative ESG screening criteria is needed to evaluate not only financial aspects but also sustainable and ethical opportunities in project delivery models. This is particularly important to consider when screening a P3 model, which is commonly chosen for delivering social infrastructure projects in Canada.

#### 1.3. Scope and Objectives

The main aim for this study is to integrate ESG criteria into the procurement process to align social infrastructure project with sustainable and responsible practices. The sub-objectives of the study are as follows:

- 1. Identify and describe ESG qualitative criteria for the procurement process of new social infrastructure projects.
- 2. Develop an ESG-PPP screening assessment matrix to evaluate the suitability of P3 models to deliver social infrastructure projects sustainably and responsibly.

5

 Develop an Analytical Hierarchy Process (AHP)-PPP selection model to guide the selection of most suitable type of P3 model to deliver a given social infrastructure project.

By achieving these objectives, this study intends to provide useful knowledge for the current state of project delivery model selection, promote sustainable practices, and provide decision-makers an opportunity to prioritize non-financial aspects of a social infrastructure project.

#### 1.4. Thesis Organization

This research comprises four chapters. Chapter two addresses current issues in social infrastructure projects, including project objectives, ESG procurement considerations, and P3 procurement models. Chapter three details how the matrix evaluates P3 model suitability for responsible sustainability and explains the AHP selection process steps in choosing a specific P3 model. Chapter four describe the practical application of the AHP-PPP selection model through case studies to assess its real-world effectiveness and reliability. Chapter five summarizes research findings on integrating ESG criteria into the screening matrix and selection model implementation. It explores implications and outlines recommendations for practical use and future research.

# **Chapter 2: Literature Review**

#### 2.1. Introduction

This chapter is comprised of five sections. The first section 2.2 will review the proposed social infrastructure projects, their characteristics, project drivers and challenges surrounding their proposed delivery in Canada. Following this, section 2.3 will explore various P3 models and the procurement process of P3 projects, as well as describe the project management and governance structure of the P3. Section 2.4. will review and identify the screening criteria required for sustainable and responsible project management of P3 projects. It will highlight existing state and key features of sustainable and responsible project management practices within the Canadian context. Finally, section 2.5 will summarise identified research gaps which will be addressed further in the methodology of this thesis.

## 2.2. Objectives of social infrastructure projects in Canada

The decision to select a specific delivery method is dependent upon the project's characteristics and objectives as well as its ability to deliver the project and other relevant considerations [20]. Thus, understanding the key objectives of the social infrastructure projects in Canada will provide a basis for the up-to-date evaluative criteria when deciding on a project delivery model. The following sub-sections of affordable housing, schools, hospital, public transit and recreation and culture facilities will provide an overview of the different types of social infrastructures, outlining their scope and commitment to addressing the various needs of Canadian communities.

#### 2.2.1. Affordable housing in Canada

Affordable housing projects include planning, construction, renovation, or expansion works focused on providing affordable and subsidized housing options for the community. Social housing is a subset of affordable housing specifically addressing the needs of vulnerable low-income populations [21]. The driving factors for these projects include an increase in housing demand

8

due to a limited supply. Canada is currently implementing a range of housing projects to address the shortage of housing options, with the goal of providing sufficient availability of affordable housing by 2030 [22]. Most provinces are actively looking for proposals to enhance affordable housing options. They are particularly focusing on creating sustainable and cost-effective housing developments as outlined in their infrastructure plans [23] [24].

Ontario is currently developing proposals to integrate housing with public transit infrastructure. These housing complexes will include commercial and communal spaces, strategically positioned around its four main transit subways, forming what is known as a transit-oriented development [25]. British Columbia is actively involved in developing several affordable housing infrastructure projects under its homes for people action plan. This plan targets the increasing demand for affordable housing among Indigenous peoples, the elderly, students, families, and individuals affected by domestic violence [26]. Saskatchewan intends to build, rehabilitate, and adapt existing social housing, and make community housing more sustainable [27]. Likewise, other provinces including Quebec, Alberta, Manitoba, and the Atlantic provinces are proposing similar approaches to address the housing scarcity and achieve Canada's objective of providing affordable housing by 2030.

The Canadian housing sector's main goal is to offer a variety of housing options that are sustainable, affordable, and quickly built, while addressing social inequalities among various community groups. However, delivering these large-scale housing projects which involves various stakeholders will pose challenges, such as a tight schedule and community opposition. Table 2 below summarises the key objectives for social housing in Canada and the ESG issues to consider.

9

Existing project drivers	Environment, Social and Governance issues to consider	
Schedule: Of fast-track nature	Social: Fast-track projects may limit the engagement of a broader range of local community groups, potentially resulting in disruptions and conflicts if community concerns are not properly addressed.	
Scope/Scale: Mix	Environmental: Large-scale projects may have significant	
housing options and	environmental impacts, such as habitat destruction or increased	
complexes	energy consumption.	
Diverse stakeholders/	Social: Effective community engagement can address social	
Community engagement	acceptance issues and ensure that the project benefits all	
	stakeholders, including marginalized communities.	
	Governance: Involving diverse stakeholders enhances	
	transparency and accountability in project decision-making	
	processes, promoting good governance practices.	
Potential risk:	Community Risks:	
	<ul> <li>Challenges related to social acceptance, such as social</li> </ul>	
	exclusion and housing inequities [28],	
	<ul> <li>Changes in neighborhood dynamics including concerns</li> </ul>	
	related to traffic, or changes in neighborhood character	
	[29]. Leading to impact on the marketability of housing	
	projects.	

Table 2. Key objectives for Social Housing in Canada.

### 2.2.1.1. Schools

School infrastructure projects include planning, construction, renovation, or expansion works focused on educational facilities. New educational facilities are built to accommodate growing student populations, replace outdated facilities, and meet specific educational requirements. The objectives of these projects include providing conducive and innovative learning environments that cater to the evolving needs of students [30], community development, and the achievement of broader societal sustainable development goals. In Alberta, similar to other provinces, the driving factors of school projects are the condition of existing school buildings, the well-being and safety of students, and enrollment trend [31]. Provincial governments are focused on building, renovating, and expanding schools to ensure equal opportunities for students and support community growth and educational excellence [22] [32] .

Likewise, there is a significant demand for building, repairing, and maintaining schools on reserve land [33]. Fostering partnerships and collaborative decision-making with Indigenous communities and tribal councils during school projects is crucial for providing culturally relevant facilities, advancing the quality of life for Indigenous people, and ensuring transparency and accountability in project planning and implementation.

During the operation of school buildings, various school programs promote sustainability practices during operation of schools, with some specifically evaluating energy and water use efficiency, waste, and green gas emission of the buildings [34]. A report highlights that schools with effective maintenance management practices achieve higher energy efficiency levels [35]. Therefore, incorporating feedback loops into a school project delivery model allows data collected from assessing waste, energy, and water efficiency in existing school buildings to inform the design and construction of new school infrastructure. Table 3 below summarises the key objectives for schools in Canada and the ESG issues to consider.

Existing project drivers	Environment, Social and Governance issues to consider
Enrollment Trends,	Environmental: Use of innovative construction techniques expertise
Educational Needs and	in the school design and construction to cater to diverse student
Innovation	needs.
	Governance: Effective coordination among stakeholders to address
	educational needs efficiently and transparently.
Indigenous needs	Environmental: Respect for Indigenous lands and ecosystems,
	ensuring construction activities do not disrupt sacred sites or natural
	habitats [36].
	Social: Respect for Indigenous rights and cultures, involving
	Indigenous communities in decision-making processes regarding
	indigenous school project implementation.
	Governance: Establishing partnerships and agreements with
	Indigenous communities to ensure their voices are heard and their
	needs are addressed throughout the project lifecycle [37].

Table 3. Key objectives for Schools in Canada.

Existing project drivers	Environment, Social and Governance issues to consider
Operation efficiency data and stakeholder involvement	Environmental: Efficient operation and maintenance practices to minimize resource consumption and environmental impacts. Social: Involvement of stakeholders such as school boards, parents, and teachers in decision-making processes related to operation and maintenance [38]. Governance: Establishing stakeholder engagement and feedback collection to improve operational efficiency and address concerns quickly
Potential Risk	<ul> <li>Operational risks:</li> <li>Challenges may arise during the maintenance and operation of P3 school projects. As some private project companies may not involve the school boards, parents, teachers, and other relevant stakeholders and their control may pose operational and maintenance restrictions [38].</li> </ul>

## 2.2.1.2. Hospitals

Hospital infrastructure projects include planning, designing, constructing, renovating, expanding, or development works that aim to build or improve healthcare facilities. The new hospital structures are built to accommodate growing healthcare needs, population increases, and replace outdated facilities. The main drivers for these projects include healthcare demands, medical technology advancements, and community health priorities. Also, the growing population and indigenous communities play a key role driving the decision-making process for hospital projects, particularly post-COVID-19 [39]. Hence, it is important to build hospital infrastructure and expand service capacity to enhance healthcare accessibility and meet the growing healthcare demands.

Although the positive objectives of these projects are evident, Canadian hospitals are recognized for their significant environmental impact, mainly attributed to the continuous and essential 24/7 healthcare services they provide. They greatly contribute to a significant environmental impact by generating waste, consuming energy, and depleting resources [40]. Because of this, projects involving hospitals need to go through a thorough planning and design

process, with an emphasis on being environmentally friendly. The primary objective for upcoming projects is to construct hospital infrastructure that not only addresses present community healthcare requirements but also prioritizes long-term environmental sustainability by incorporating eco-friendly practices throughout the entire project lifecycle.

Likewise, hospitals are typically built to last 30-40 years, are struggling with delayed maintenance in Canada [41]. Essential renovations are being postponed due to funding shortages in the public sector [42]. In order to address this problem, it is important to integrate maintenance and operations planning into the design and delivery of new hospital facilities. Additionally, involving maintenance personnel is valuable as they can share their expertise on the functioning of existing projects, pinpointing areas for improvement and helping avoid potential issues [43]. Table 4 below summarises the key objectives for hospitals in Canada and the ESG issues to consider.

Table 4. Key objectives for Hospitals in Canada.

Existing project drivers	Environment, Social and Governance issues to consider
Scope/Scale: New	Environmental impact concerns: Consider involving private sector
construction of	expertise and innovation in green building processes and
specialised facilities	technologies. Addressing concerns regarding the hospital project's
	overall environmental footprint, including energy consumption,
	waste generation, and carbon emissions.
	Social: Consider the community's response to the construction of
	a new specialized facilities, including disruption to local
	businesses or residents.
Funding: Budget	Governance: Ensure transparency and accountability in budget
constraints and value-	allocation and expenditure and possibility of early contractor
for-money	involvement, particularly regarding environmental and social cost
	management.
Operational efficiency:	Environmental: Long-term sustainable practices in the operation of
Environmental operation	specialized facilities.
footprint	

Existing project drivers	Environment, Social and Governance issues to consider	
Stakeholder Involvement: Navigate the complexity of stakeholders and gathering diverse input	Social: Facilitating meaningful participation from diverse stakeholders, including local residents, environmental advocacy groups, and regulatory agencies in discussions about environmental concerns and potential mitigation strategies.	
Potential risk	<ul> <li>Environmental Risks:</li> <li>Contamination issues, Ecological considerations, Environmental impact concerns including challenges relating to adopting green building process [44]</li> </ul>	

## 2.2.1.3. Public Transit

Public transit infrastructure projects involve planning, designing, constructing, renovating, expanding, or developing facilities that support and enhance public transportation services within a provincial city. These include stations, rail systems (like metros or subways), and bus rapid transit lanes [45]. They serve an important role in shaping the efficiency, accessibility, and sustainability of public transportation systems, contributing to the overall mobility and livability of urban areas [46]. The factors driving these projects include changes in transportation needs, population growth, technological advancements, and government priorities. Canada is dedicated to creating healthier, sustainable transit and increasing public transportation usage for their daily commutes [46] [47]. However, these projects necessitate substantial financial commitment and coordination. Currently, they face an operational funding shortage, putting their ability to effectively serve communities at risk [48]. Likewise, specific projects, such as Ottawa's Light Rapid Transit, Edmonton's Light Rapid Transit line, and Toronto's Eglinton Crosstown LRT line, encountered setbacks, prolonged schedules, and budget overruns during its construction, primarily due to funding-related challenges [47]. Subsequently, securing lifecycle reliable and sustained lifecycle funding is an important consideration for the successful delivery and completion of public transit projects.

On the other hand, equity-deserving groups, such as racialized people, Indigenous communities, persons with disabilities, newcomers, seniors, low-income individuals, and people

14

experiencing homelessness, are recognized as the primary users of public transit [47]. To guarantee effective solutions that meet their community transit needs, it is important to acknowledge and address the social and environmental challenges faced by these community groups early on, during the planning and implementation phases. Similarly, improved collaboration and communication among different levels of government is important. Implementing a transparent decision-making process grounded in evidence is vital for the development and delivery of transit projects, aiming to improve project governance [49]. Table 5 below summarises the key objectives for public transit in Canada and the ESG issues to consider.

#### Table 5. Key objectives for public transit in Canada

Existing project drivers	Environment, Social and Governance issues to consider
Scope/ Scale: Deliver diverse range of transit options.	Social: Disruption to local communities and displacement of residents due to construction and transit infrastructure expansion.
Operational funding: Reliable and sustained lifecycle funding	Environmental/Social: Green funding sources to support ongoing operations and maintenance. Green funding refers to financial mechanisms specifically designed to support projects that have positive environmental impacts or contribute to sustainability goals [50].
Community engagement: Transparency, coordination, and evidence-based decision making	Social: Promoting inclusivity and addressing the needs of marginalized communities in transit planning and implementation.
Potential Risk	<ul> <li>Operational and Maintenance Challenges:</li> <li>Delays, reduced scope, or project abandonment due to funding shortages. Also, delivered transit options overestimation or underestimation of future ridership, impacting revenue and financial sustainability for the project.</li> </ul>

## 2.2.1.4. Recreation and Culture infrastructure

Recreation infrastructure projects involve planning, designing, constructing, renovating, expanding, or developing facilities and spaces that support and facilitate recreational activities and leisure pursuits. These infrastructures include museums, indigenous heritage centers, sports, and aquatic facilities and other indoor and outdoor recreational community centres [2] [51]. They serve

an important role in advancing Canada's social-cultural objectives to provide diverse opportunities for individuals of all ages and abilities to engage in leisure activities, promote community wellbeing, and create a sense of place and belonging. Delivering these projects necessitates addressing diverse demographic needs in a community [52].

Likewise, with the growing number of immigrants and increasing ethnocultural diversity in various communities, a key challenge identified in the delivery of culture and recreation infrastructure in Canada is the rise of changing cultural practices [53]. This poses a great risk of project acceptability. Addressing this challenge involves anticipating future changes in culture, demographics, and the economy, emphasizing the need for a forward-looking approach in the planning and delivery of culture and recreation infrastructures. On the other hand, Insufficient funding for maintenance and operation by the municipality poses challenges, impacting the sustainability and longevity of the facilities [53]. Table 6 below summarises the key objectives for recreation and culture in Canada and the ESG issues to consider.

Existing project drivers	Environment, Social and Governance issues to consider		
Stakeholder diversity:	Governance: Ensuring all stakeholders are engaged throughout		
Transparency and	the project lifecycle, with transparent communication and decision-		
inclusivity	making processes.		
Post Construction	Environment: Considering the long-term sustainability of		
Phase: Maintenance	maintenance practices to ensure environmentally safe and long-		
consideration	lasting infrastructure assets.		
Potential Risk	Community Opposition:		
	<ul> <li>Resistance or opposition from the community due to</li> </ul>		
	concerns over project impact, design, or perceived lack of		
	consultation		

Table 6. Key obje	ectives for r	ecreation and	culture in	Canada
-------------------	---------------	---------------	------------	--------

In summary, new social infrastructure projects phases need to be carefully coordinated to tackle identified ESG considerations and potential risk in delivering sustainable infrastructure project outcomes. Each project phase must address environmental responsibility, social equity, and transparent decision-making for the community. These considerations will influence how the

project is planned, designed, and managed [54] [55]. Currently Canada plans, designs, and manages social infrastructure projects using either a traditional procurement, where the government fully funds projects internally or a P3 procurement, which involves the private sector participating in funding and delivering the project [56]. Table 7 below shows the different categorises of PDM used in Canada for social infrastructures.

Category of PDM used in Canada for social infrastructures	PDM Types
Sequential model	Design-Bid-Build (DBB)
Overlap models	Design-Build (DB),
	Design-Build-Finance (DBF)
Innovative models	P3 – DBFM, DBFOM
	P3 Bundle
Collaborative models	IPD/ Alliance
	"Progressive" P3

Table 7. Category of PDM used in Canada for social infrastructures

## 2.3. Public-private partnership models in Canada

P3s are collaborative arrangements between the private sector and government to manage various phases of an infrastructure project, including design, construction, funding, oversight, and maintenance. These projects phases are consolidated into a single contract, spanning 25 to 30 years [57], comprising the core services to be delivered by the private sector [58]. The typical structure of a P3 model in Canada constitutes two main parties; the government who is the owner and the project consortium which is comprised of number of stakeholders acting as one private entity [56] as seen in Figure 3 below. Likewise, Figure 4 below shows the basis for the variation of the P3 models in Canada; it visualizes how traditional and P3 models vary in terms of private sector involvement and risk transfer. The PDM situated towards the upper right involve significant private sector participation and more-risk transfer, while those towards the lower left represent PDM with lower private sector involvement and less risk transfer [57]. Table 8 below outlines the main characteristics of P3 models in relation to their scope of core and non-core services.



Figure 3. Partnership Structure in a Typical P3 Model in Canada





Type of P3 model	Core Services	Non-Core services (supportive of the core services)
DBFM	Focuses on design, construction,	1. Cleaning and janitorial
	financing, and maintenance aspects.	services
DBFOM	Extends beyond maintenance to	2. Security services
	operational aspects.	3. Landscaping and grounds
P3 bundle model	Involves bundling smaller scale	maintenance
	projects into one large project	<ol><li>Waste management and</li></ol>
Progressive P3 model	Engaging private stakeholders, including contractors, at an early stage to collaboratively define project specifications and pricing before finalizing the project contract [59].	<ul> <li>recycling</li> <li>5. Catering and food services</li> <li>6. Facility management and building maintenance</li> <li>7. IT support and helpdesk services</li> <li>8. Administrative and clerical support</li> <li>9. Transportation and shuttle services</li> </ul>
		10. Event planning and coordination

Table 8. Types of P3 models in Canada

Non-core services may be required throughout various stages of the project lifecycle, including during construction, operation, and maintenance phases. However, they are typically provided on an ongoing basis to ensure the smooth functioning and upkeep of the project's facilities or operations [58] [60].

Different provinces use different P3 models for certain social infrastructure projects. In Ontario, the DBFM model has been extensively procured for most of the hospital projects currently in their operational phase [61], as depicted in the Figure 5 below. Additionally, a new model called the progressive P3 model was introduced in 2021, which has been predominantly involved in recent transit projects in the province. Unlike traditional P3 models, the progressive P3 model involves private stakeholders in the early planning phases of the project before finalizing the draft project agreement [59]. This approach enables a collaborative review of project objectives and risk factors, ensuring a shared understanding among stakeholders [10].



Figure 5. Ontario's P3 models across Different Social Infrastructure Projects

In Alberta and Saskatchewan, a variation of the P3 model known as the P3 Bundle is used for constructing schools, as depicted in Figure 6 and 7 below. The Alberta schools' phase 1,2,3 has been delivered using a single bundled DBFM contract, with maintenance periods extending until 2040, 2042, and 2044 respectively [62]. Likewise, the Saskatchewan joint use school projects were constructed using a 32-year bundled DBFM model with maintenance periods extending to 2045 [63]. However, concerns have been raised about the suitability of the DBFM P3 model for these projects. In Alberta, criticism is focused on limited school administration control and delays in addressing maintenance issues [64]. Similarly, in Saskatchewan, the school projects have faced criticism for significantly higher maintenance costs compared to older non-P3 school projects [65]. Other provinces like Quebec, British Columbia, and New Brunswick primarily use P3 models for hospitals as seen in Figure 8 below, and provinces like Yukon and Prince Edward Island do not have any presence of P3 social infrastructure projects, as seen in Figure 9 below.



Figure 6. Alberta's P3 models across Different Social Infrastructure Projects



Figure 7. Saskatchewan's P3 models across Different Social Infrastructure Projects



Figure 8. P3 models across different Social Infrastructure Projects: Québec, British Columbia, New Brunswick, and Manitoba [9]



Figure 9. P3 models across Different Social Infrastructure Projects: NL, Nova Scotia, PEI, and Yukon [9]

## 2.3.1. Procurement of a P3 model for social infrastructure in Canada

The procurement of PDMs occurs during the project's planning phase, as shown in Figure 10 below. It begins with identifying and reviewing project scope elements, key project risks, and available PDM options. Afterward, procurement objectives are established along with a set of defined criteria to guide the evaluation of the most suitable procurement options for project delivery. A thorough qualitative assessment is then conducted to rank the available procurement options based on these established criteria [66]. This qualitative stage results in a shortlisted set of PDMs and involves evaluating the potential for the project to be procured as a PPP. After, a second stage involves quantitative analysis, also known as VfM compares the shortlisted PPP model with the traditional PDM option. It calculates the risk-adjusted project costs to compare financial advantages expressed as cost savings benefits [54]. Figure 11 below illustrates these stages in the P3 procurement process. If the value for money is negative for the PPP model, a review of the qualitative analysis results is conducted, and consideration is given to procuring a non-PPP model as seen Figure 12 below. Thus, it is important to recognise that while P3s may be suitable for numerous project scenarios, they may not always be the most suitable option for small project sizes with no significant risk to the public sector [67] [58].



#### Figure 10. Procurement Phases of a Social Infrastructure P3 Model in Canada



Figure 11. The Two-Stage Procurement Process of a P3 Model in Canada



Figure 12. Existing evaluation framework for assessing P3 model potential in Canada [68]

The qualitative assessment relies on guidelines provided by P3 Canada and a Triage tool from PWGSC's Strategic Sourcing Sector in 2021 [68] [56] [69] to develop qualitative screening matrices. These guidelines provide a systematic process for screening infrastructure projects for P3 potential. Provinces across Canada customize qualitative assessment matrices to align with their infrastructure needs and provincial government priorities when evaluating the suitability of a P3 model for infrastructure procurement. These matrices incorporate various evaluative criteria, including private sector interest, project characteristics, risk transfer opportunities, innovation, and financial considerations [56] [70]. However, the decision to pursue P3s is guided by shared specific circumstances and objectives across provinces. Existing factors such as project size, complexity, risk, and funding availability are significant influencers in the procurement decision-making process. This study has compiled common key scenarios and identified thresholds across provinces like Ontario, Alberta, British Columbia, Quebec, and Saskatchewan regarding the use of P3s for social infrastructure projects described in Table 9 and illustrated in Figure 13 below, providing an understanding of the existing key assessment criteria under which provinces in Canada have considered the P3 model for social infrastructure project procurement. However, these considerations often overlook important environmental, social, and governance factors, which are important for assessing the sustainability and responsible performance of P3 projects. Neglecting these aspects may result in an incomplete evaluation of the project's full value, including its impact on people and the environment [71] [72].

Project	Alberta	British	Ontario	Quebec	Saskatchewan
			ernane,	<b>Q</b> (10)000,	edenatorieriari
Criteria		Columbia,			
Project size	Project	size exceedir [7	ng 100 millio '5]	on [73] [74]	Projects with a capital cost of \$50 million or more may undergo screening for potential P3 viability, provided there is an inclusion of maintenance and/or operations components and has significant risk [76]

Table 9. Common Factors and Key	y Thresholds for Considering P3s in Canadian P	'rovinces				
	· · · · · ·					
Project	Alberta,	British	Ontario,	Quebec,	Saskatchewan	
---	---	--	---	--	--	--
Criteria		Columbia,				
Value for Money	When pc both qu The lo off	sitive VfM is uantitative an west bid may er adequate	achieved by d qualitative not win if it value for mo	v evaluating benefits. does not oney.	If the VfM exceeds 3%, P3 consideration is pursued. However, if it falls below this threshold, a sensitivity analysis is conducted to determine the feasibility of a P3 model [76]	
Competitive and Fair Procurement Process	Availability of three project consortiums to bid for a project [77]. This has been a rising challenge as risk uncertainties to the private sector has limited the bidding pool for P3 model procurement [78].					
Innovative Design Delivery and Risk Allocation	Most soc integratic of cost sa partners	ial infrastruc on opted to se avings from i [79]	ture projects creen the pro nnovative so	with comple ojects for P3 olutions and	ex designs and technology procurement to gain benefits risk transfer provided by P3	



Figure 13 Existing P3 Decision Flow Chart for Social Infrastructure Projects in Canada

Various VfM reports highlights differing cost-saving benefits across different P3 models [80]. While reports for models like DBFM and DBFOM typically range from 15% to 22% [81], VfM assessments for progressive P3 models are currently unavailable due to limited projects reaching financial close. However, early stakeholder engagement in progressive models is expected to enhance VfM considerations [11]. This proactive approach allows for the early identification of factors impacting financial implications and broadens the scope of VfM scenarios. This timely stakeholder involvement in the planning process has the potential to enhance project accountability, transparency, and performance expectations. Purposely, this could improve overall contract integrity and facilitate efficient P3 project governance [70].

### 2.3.2. PPP project governance in Canada

Governance is defined as a structured system of decision making that ensures effective coordination, accountability, and transparency throughout the project lifecycle [82]. This ensures successful delivery of project objectives and meets stakeholder interests [83]. When decision-making lacks transparency, it often results in cost overruns and budgeting discrepancies. Similarly, weak governance structures can contribute to a lack of accountability, which hinders the timely identification and resolution of issues [82]. As a result, this can lead to delays and failures in delivering infrastructure projects. The P3 governance structure includes the organizational framework established to oversee and manage a P3 project. It outlines the roles, responsibilities, and decision-making processes of each stakeholder involved, including government agencies, private sector partners, and possibly other entities such as regulatory bodies or community council representatives [84]. Different governance structures comprise different project governance mechanisms. Governance mechanisms refer to the different undertakings within the organizational framework put in place to oversee and manage project stakeholders, communication channels, risk management procedures, and performance monitoring schedules [85]. Canada's

P3 governance structure integrates four governance mechanisms, as described in Table 10 below, to ensure that the P3 projects are executed according to established objectives, schedules, and budgets, while also addressing any potential risks or issues that may arise during project implementation.

Governance mechanisms type	Scope of definition	Governance mechanisms used (Formal)
Contractual governance	This is a legal agreement which defines the rights, responsibilities, and obligations of each party, as well as the terms and conditions between the public and private sectors involved in a P3 project.	<ul> <li>P3 project agreement adhering to the IPFP Framework – Ontario [86]</li> <li>P3 project agreement including a Tri-party agreement – Alberta School projects [87]</li> <li>Lenders' Direct Agreement</li> <li>Independent Certifier Agreement</li> </ul>
Relational governance	This mechanism focuses on the interpersonal relationships, communication channels, and collaboration frameworks established between the public and private sectors, and the community. It aims to foster trust, cooperation, and mutual understanding among project stakeholders to ensure effective project delivery. It may involve informal processes, such as meetings, negotiations, and relationship-building activities, as well as collaborative problem- solving in response to unexpected occurrences [88].	<ul> <li>Community Benefits Framework         <ul> <li>Community benefits agreement</li> <li>Community Engagement and Stakeholder Relations Plan [86]</li> </ul> </li> </ul>

 Table 10. Government Mechanisms and Considerations in Canadian PPPs

Governance mechanisms type Risk governance	Scope of definition This mechanism is designed to identify, assess, mitigate, and	Governance mechanisms used (Formal) Risk identification and resolution clauses in project agreement contract
	manage risks associated with P3 projects. It includes strategies on identification of partnership risks, procedures to communicate identified risks, and decision- making steps to minimize potential negative impacts on project outcomes.	<ul> <li>forms [86]:</li> <li>Proceeding at Risk Matters and Notice</li> <li>Dispute resolution procedure</li> <li>Force majeure</li> </ul>
Reporting governance	This mechanism refers to the structured process by which information is collected, documented, analyzed, and communicated within a project. It ensures reporting on progress, achievements, and challenges of the project, facilitating effective communication.	<ul> <li>A tracking and reporting system to demonstrate the progress of:</li> <li>The Community Benefits and Liaison Plan</li> <li>Construction Progress Schedules</li> <li>Financial performance and cost budgets including change orders.</li> <li>Quality management and the Monthly Performance Monitoring Reports in adherence to the performance monitoring program – Ontario [86]</li> <li>Energy performance monitoring</li> </ul>
Environmental governance	I his mechanism refers to the process involving setting standards and requirements to protect and preserve the environment. It helps regulate environmental issues by promoting sustainable practices and reducing environmental risks	<ul> <li>LEED Rating</li> <li>Toronto Green Standard Design and Construction Obligations [86]</li> </ul>

As seen in Table 10 above, the P3 project agreement is a contract governance mechanism that ensures effective management [89]. In Canada, the project agreement also provides and describes the undertakings of relational, risk, and reporting governance mechanisms. The CCDC

and impacts of a P3 project.

(Canadian Construction Documents Committee) contract forms are widely acknowledged and adopted as the standard references for governance frameworks used in the execution of largescale social infrastructure projects. Most social infrastructure projects in Canada are governed by either a CCDC 2 for Design-bid-build delivery or CCDC 14 for Design-build or P3 project delivery. These contract forms are highly regarded for their perceived objectiveness and fairness in ensuring effective project delivery [70] [90]. However, reports of transparency issues and corruption in P3 social infrastructure projects [16] indicate a lack of emphasis on fulfilling the contract roles with accountability. Thus, to mitigate these transparency and accountability issues, monitoring and evaluating stakeholder management is essential.

Relational governance involves stakeholder management in the organizational framework. During the planning and procurement stage, the P3 project team is formed, often including thirdparty advisors [57] [91]. In Canada, common advisors for P3 projects typically include legal, technical, and financial experts. However, it remains uncertain whether environmental and social advisors are deemed essential as part of the technical advisory team. The environmental and social advisors may play an important role in P3 project implementation for several reasons as firstly, they ensure compliance with environmental regulations and social responsibilities, minimizing negative impacts on the environment and communities. Their expertise helps identify potential environmental risks and social concerns early in the project lifecycle, allowing for mitigation measures. Additionally, environmental and social advisors contribute to stakeholder engagement efforts, facilitating communication with local communities and addressing their concerns [92]. Overall, their involvement for relation governance promotes sustainable development, enhances project credibility, and reduces the risk of delays or conflicts arising from environmental or social issues [92]. Hence, understanding stakeholder changing aspects in P3 projects is important for sustainable development and project success [93]. Figure 14 below illustrates the key stakeholders that form the key components of a relation governance structure

in Canada. The community stakeholders ar among the key stakeholders that have an impact to P3 projects and are broadly impacted by the P3 project [92].



Figure 14. Stakeholder Structure in P3s in Canada [57]

Public Community Benefits Agreement (CBA) have been employed to ensure effective relation governance of community stakeholders in P3 projects. A CBA is a is a contractual agreement that specifies and upholds the advantages that a community gains from a specific infrastructure or development project. It particularly focuses on marginalized groups vulnerable to social exclusion, discrimination, poverty, and violence, ensuring their inclusion and well-being within the project's benefits. These marginalized groups include recent immigrants, persons with disabilities, women, veterans, youth, Indigenous peoples, and local social trade enterprises. The CBA aims to establish fundamental standards for offering employment, training, and community supply chain opportunities to marginalized groups [94]. Despite their potential, public CBAs have been underutilized in social infrastructure projects within P3s,particularly for hospitals as illustrated in Figure 15 below [94] [95]. However, with Canada's commitment to fostering more benefits to inclusive communities through the Investing in Canada Infrastructure Plan, CBAs are expected to become more common in social infrastructure projects [2].



Figure 15. Community Benefit Agreements in Canadian Social Infrastructure Projects

The environmental governance mechanism in Canada often relies on third-party certifications, commonly utilized to ensure the environmental sustainability of projects. Third-party certifications, such as LEED, are standardized rating systems used to evaluate the environmental performance and sustainability of buildings, infrastructure, and other projects. These certifications are typically awarded by CAGBC and assess various aspects such as energy efficiency, water conservation, materials selection, indoor environmental quality, and sustainable site development [96]. LEED is adopted during the early planning stages to ensure social infrastructure buildings are designed, built, and operated to ensure high-performance, cost-effective project outcomes, ensuring they meet high standards of sustainability and environmental responsibility. Other examples of third-party certifications include TRUE which evaluate buildings for zero waste performance and

ZCB which ensures low-carbon design and operational performance for buildings. Their use in social infrastructure projects is seen in the Figure 16 below [97]





Out of the 3,316 building projects certified under LEED, only 158 hospitals, 360 education facilities, and 202 recreation and culture centres got LEED certification by 2024. Furthermore, only 29 public transit passenger station projects obtained LEED certification. Offices and mixed-use offices had the highest number of certifications, with a total of 1,096 projects [98]. This indicates that social infrastructure projects, particularly in areas like hospitals, education, recreation, and public transport stations, are falling behind in the adoption of environmentally friendly certifications compared to other sectors such as offices and mixed-use spaces. Likewise, out of 144 projects certified with the ZCB standard, only 10 were community and sports facilities, and 5 were education projects. No hospitals or public transit projects received certification. Similarly, the TRUE standard, aiming for zero waste for infrastructure and building projects, has certified 12 projects, none of which are social infrastructure projects covered in this study. Hence, LEED has

been predominantly used for social infrastructure projects as seen in the Figure 16 above and school projects in Figure 17 below.



#### Figure 17. Third-Party Certification in Canadian Infrastructure Projects

The effectiveness of a P3's organizational structure relies on the governance mechanisms put in place, as discussed earlier. These mechanisms directly impact the project's ability to achieve its goals [85]. The above mechanism show the different frameworks addressing governance, environmental, and social aspects, collectively addressing them as ESG issues. The adoption of social and environmental governance mechanisms is still in its early stages and is not active across all project phases. However, recently, the CCPPP introduced a national award category to evaluate projects based on various ESG considerations. This award category could encourage more P3 projects stakeholders to have an incentive for the integration of Environmental and Social governance mechanisms in P3 projects, particularly from the planning phase of P3 procurement, to assess the project's contribution to environmental protection, wildlife protection, climate/GHG

impact, natural infrastructure attributes, Indigenous engagement, community benefits, and other social impact [99]. Hence, ESG governance mechanisms can collectively contribute to the improvement of sustainable and responsible project management for social infrastructure projects in Canada.

## 2.4. Sustainable and Responsible P3 project management

Sustainable project management refers to the process of minimizing negative impacts and maximize positive outcomes and guarantee the long-term success of projects [14]. Sustainable project management involves examining sustainability factors and how they impact various aspects of projects [100]. Responsible project management emphasizes ethical, social, and environmental considerations throughout the project lifecycle [101]. It prioritizes transparency and accountability among stakeholders, ensuring adherence to ethical practices. The main objective is to optimize stakeholder value by ensuring the successful delivery of infrastructure projects while making ethically sound decisions [102] [103]. In the context of P3 projects, there is a growing recognition of the importance of incorporating sustainability principles factors into the project management process to ensure that projects are developed and managed in a sustainable and responsible manner [104]. Integrating sustainability considerations into project management process is an important step towards achieving long-term sustainability value [105]. It considers how sustainability influences the project opportunities available, the value that can be added, and the challenges that may arise throughout the project management process [106] . This may involve conducting sustainability assessments, defining sustainability criteria for project delivery, engaging stakeholders, and promoting transparency and accountability throughout the project lifecycle as part of project management governance mechanism [107] [108]. Table 11 below provides a summary of existing findings and recommendations for sustainable and responsible project management, including insights on the integration of ESG considerations.

No	Title	Source	Year	Findings	Gap/Recommendations
1.	Sustainable Project Management: Trends and Alignment	Journal Article	2020	Identified the integration of sustainable practices into project management not only enhances stakeholder value but also mitigates project risks. Additionally, it plays a key role in overall project organizational success within the UAE.	The study lacks specific discussions on sustainability considerations tailored to particular types of infrastructure projects. Additionally, responsible project management has not been adequately addressed.
2.	How project management practices lead to infrastructure sustainable success: an empirical study based on goal- setting theory.	Journal Article	2020	The specific culture, strategy, implementation, and reflection of project management practices in Chinese infrastructure projects were justified, demonstrating their impact on the attainment of infrastructure sustainable success.	More case studies can be investigated to strengthen the correlation between project management practices and the sustainable success of infrastructure projects

Table 11. Findings and Recommendations for Sustainable and Responsible PM

No	Title	Source	Year	Findings	Gap/Recommendations
3.	A systematic literature review concerning the different interpretations of the role of sustainability in project management	Journal Article	2021	The researchers found three main perspectives on sustainability in project management: seeing it as a constraint, as a tool to support project goals, and as intrinsic value, forming a business case for sustainability. They also introduced a new definition of sustainable project management, emphasizing a value- oriented approach.	Further research is needed to explore the impact of various project delivery models on integrating sustainability into project management, utilizing empirical methods like case studies. Additionally, there is a need for research in defining the conditions and methods for considering weak and strong sustainability in project management.
4.	Barriers to sustainable construction project management: the case of Iran	Journal Article	2021	The researchers identified three obstacles to adopting sustainable practices in Iran's construction projects: a lack of collaboration, insufficient identification of sustainability objectives in construction projects, and a limited understanding of the benefits of sustainability.	The focus has been on identifying barriers to sustainable project management. More research should focus on the specific considerations required for effective sustainable project management.
5.	Critical Factors to Achieve Sustainability of Public-Private Partnership Projects in the Water Sector: A Stakeholder- Oriented Network Perspective	Journal Article	2021	The study highlights specific factors and challenges in the sustainability of water P3 projects. It delves into factors like ecological awareness, construction quality, ecological designs, and project management capacity that significantly influence the sustainability of such projects.	Future research opportunities in this area could involve exploring additional stakeholder perspectives and focusing deeper into the role of financial investors in influencing project sustainability.

No	Title	Source	Year	Findings	Gap/Recommendations
6.	Building bridges: Unraveling the missing links between Public- Private Partnerships and sustainable development	Journal Article	2022	The study identified a strong link between P3s and the attainment of UN Sustainable Development Goals. It emphasized significant interdependencies between P3s and their potential contributions to social, environmental, and economic aspects.	Further research in the context of P3s, including assessing the environmental aspects of P3 projects and the social acceptance of P3s as a sustainable project model in different contexts.
7.	Investigating the significance of sustainability indicators for promoting sustainable construction project management	Journal Article	2023	The study identified and ranked key indicators for sustainable project management in construction projects. Using the relative importance index, revealed that environmental indicators are considered the most important.	Explore specific project management strategies and their role in achieving sustainability goals in construction projects. Investigate how improved communication channels and practices among stakeholders can positively influence social and ethical aspects of sustainability in project management.
8.	Sustainable Public procurement of infrastructure and Human Rights	Book	2023	It emphasizes the importance of sustainable procurement practices from various perspectives, Through case studies and analysis across social infrastructures sectors such as hospitals and, recreation and sports centres. It addresses challenges faced by different stakeholders, including communities, workers, and service users, proposing solutions to advance the sustainable public procurement agenda	Provides challenges and aspects that give more information on best way to promote infrastructure development that prioritizes sustainability and stakeholder rights.

No	Title	Source	Year	Findings	Gap/Recommendations
9.	Public–private partnerships: a collaborative framework for ensuring project sustainable operations	Journal article	2024	The research focused on understanding factors influencing the operational sustainability of P3s. Using Social Network Analysis and Interpretative Structure Model, it identified weak government leadership and institutional environments as key factors. These factors introduce uncertainties, delays, and inefficiencies during project implementation, impacting the long-term sustainability of P3s.	The study's findings give practical guidance for P3 project management for sustainability considering hindering factors. More research can go into other case studies.
10.	12 Responsible Leadership in Megaprojects	Book	2023	The study outlines four key levels of responsible leadership in megaprojects: Stakeholder relationships, Day-to-day operations, Leadership strategies and Personal qualities of leaders	More research is needed to examine the current role of project managers in megaprojects. This includes identifying the challenges and opportunities project managers encounter when applying responsible leadership principles in practice.
11.	Governing public- private partnerships for sustainability: An analysis of procurement and governance practices of PPP infrastructure projects	Journal Article	2017	Identified that public procurers are facing challenges in defining and measuring social sustainability criteria that can be effectively enforced. Additionally, the current contractual structure of P3s may be better suited for achieving a "weak" rather than a "strong" level of sustainability, as observed in reviewed case studies.	Further research is needed to explore the measurability and weighting of sustainability criteria in evaluating P3 projects across various case studies. Additionally, there's a necessity to review P3 governance contracts to improve the promotion of sustainability goals to be achieved.

No	Title	Source	Year	Findings	Gap/Recommendations
12.	PPP projects: improvements in stakeholder management	Journal Article	2020	Developed a comprehensive framework for managing external stakeholders, featuring dynamic stakeholder identification and consideration of their interests at each project phase.	More case study research is needed to understand stakeholder needs for effectively developing stakeholder management frameworks in P3 projects. Continuous updates are crucial as new stakeholders emerge and existing stakeholders' interests evolve.
13.	Public–Private Partnerships for Environmental, Social, and Governance Projects: How Private Funding for Infrastructure Can Produce Mutual Benefits for Companies and the Public	Journal Article	2022	Examines how P3s can be used as a tool to address ESG issues ensuring mutual benefits for different stakeholders. It focuses mostly on environmental criteria current emphasis on climate change especially in the context of the COVID- 19 pandemic.	Future studies could research deeper into the specific mechanisms through which P3s can effectively address ESG challenges.
14.	Project Sustainability and Public-Private Partnership: The Role of Government Relation Orientation and Project Governance	Journal Article	2022	It investigates how the attitude and behavior of the government towards its relationships with private sector entities influence the sustainability of P3 projects. Identifies instrumental and rent- seeking relation orientations directly impacting project sustainability. Also points the role of governance mechanisms as a facilitator between government's relation orientation and project sustainability	The study focuses just on government behavior without considering private sector behaviour and responses. Recommends future research to include private sector behavior, incorporate project characteristics, and conduct case studies for a more comprehensive understanding.

No	Title	Source	Year	Findings	Gap/Recommendations
15.	Influence of Project Governance Mechanisms on the Sustainable Development of Public-Private Partnership Projects: An Empirical Study from China	Journal Article	2023	Contractual, relational, and risk governance mechanisms positively influence P3 project governance performance which in turn positively impacts project sustainability. The study emphasizes the importance of integrating risk governance mechanisms into traditional governance systems for P3 projects.	More research on different types of P3 projects and their governance mechanisms for more detailed insights on their influence to improve project sustainability in various types of P3 projects
16.	A multicriteria decision framework for governance of P3 projects towards sustainable development	Journal Article	2023	The research found that in Ghana, contractual governance is more important for ensuring the sustainability of P3 projects compared to non-contractual factors. Effective risk allocation and communication were highlighted as key elements within contractual arrangements that significantly impact sustainability by influencing their effectiveness and implementation.	Further research should focus on expanding governance factors and investigating their influence on project success. A hierarchical decision framework was created using the DEMATEL method, involving 30 experts to evaluate two main P3 governance factors and eleven subfactors. Alternative methodologies should be considered to further explore P3 governance factors.
17.	Construction of Performance Evaluation Model for the Operation and Maintenance of Government Enterprise Cooperation Infrastructure Projects under the ESG Concept	Journal Article	2023	The study finds the positive impact of the ESG on improving the management of operation and maintenance tasks in P3 infrastructure projects.	This study suggests evaluating the effectiveness of integrating of ESG principles into project management and assessing their impact on project sustainability outcomes

No	Title	Source	Year	Findings	Gap/Recommendations
18.	A systematic review of the interrelation of administrative, Environmental, Social and Governance of Public-Private Partnerships (P3) Spaceport Project in Biak, Papua, Indonesia	Journal Article	2022	The study identifies that there is a need for a more comprehensive approach to integrating ESG considerations into the project. There is risk for challenges arising from the lack of legal clarity and the imbalance between business interests and ESG	The study suggests the importance of integrating ESG considerations throughout the project lifecycle, from planning to implementation to ensure sustainability and effectively mitigate the ESG risks, particularly in culturally and environmentally sensitive regions like Biak.
19.	Public–Private Partnerships and Green Financing of Infrastructure Projects	Part of a book series	2023	The findings emphasize the importance of sustainable and responsible investing based on ESG standards for infrastructure projects.	Future research could explore the effectiveness of green financing mechanisms in P3 projects, examining case studies and comparative studies of successful implementations in different context.
20.	Sustainable Financing for ESG Practices	Part of a book series	2024	The findings examine various sustainable financing alternatives available to organizations, including both traditional financial instruments and innovative funding mechanisms. It identifies the growing importance of sustainable financing in facilitating the implementation of ESG practices within the corporate sector.	Future research could evaluate the effectiveness of sustainable financing methods in achieving ESG goals within the construction sector, particularly in P3 infrastructure projects. Comparative studies could assess the outcomes of contexts that have implemented sustainable financing practices against those that haven't, shedding light on the advantages and obstacles of each approach.

It is clear that considering and integrating ESG aspects throughout a P3 project implementation enhances efficiency and sustainability [19] [55]. ESG issues refer to factors beyond financial metrics that are used to evaluate the sustainability and overall impact of infrastructure projects. They are important determinants of the non-financial value of a project and have been currently used to inform decision making and evaluate non-financial project

performance [19]. By assessing environmental impact, social outcomes, and governance practices, stakeholders can measure how effectively a project aligns with sustainability goals and societal needs. This ESG data assessment is usually specific and largely subjective and vary significantly based on the type of project depending on the stakeholder preferences, policies, and investment priorities [109] [110]. Thus, the early and diverse engagement of stakeholders in defining ESG issues is important to avoid stereotypes and biases. A report by Statistics Canada [111], highlights that Canada should aim to have a standardised ESG assessment guidelines among its provinces while also incorporating the unique values and interests of Indigenous peoples. This implies a recognition of the diversity of Indigenous perspectives and the importance of respecting their contributions to sustainability and governance efforts. Similarly, it is important to conduct ongoing monitoring of the environmental impact assessment for social infrastructure projects to ensure environmental concerns are addressed throughout the project lifecycle. Presently, environmental impact assessments are primarily conducted during the planning stage [112]. However, there is a lack of reported evidence on how various P3 models have monitored and addressed the identified environmental issues throughout different phases of the project implementation.

As described above, P3 projects are typically evaluated based on their potential to deliver high value-for-money, considering factors such as project size, complexity, and innovation requirements. The environmental and social governance mechanism is addressed mostly as additional components to the project agreement rather than integral parts of the project governance structure. To address this gap, the primary objective of this thesis is to identify specific ESG criteria that should be incorporated into the P3 procurement for social infrastructure projects to promote the inclusion of governance mechanisms such as CBAs in PPPs, with the aim of improving social benefits provided by PPP projects in different phases. Table 12 below presents the qualitative ESG criteria identified (see appendix I), along with their respective sources and

their application in assessing various project delivery models. These Canadian-specific ESG criteria are sourced from the identified objectives and challenges in social infrastructure projects. as described in the tables above. Additionally, they are derived from Infrastructure Canada's sustainability objectives, provincial infrastructure plans, and considerations outlined in CBAs and LEED standards. They will serve as the foundation for the proposed ESG-PPP suitability screening matrix, enabling a broader comparison with traditional project delivery models. This procurement approach aims to improve the governance framework, ensuring that projects align with Canada's sustainability goals and deliver long-term benefits to communities considering both core and non-core PPP servives. Therefore, a decision flow chart guiding when to use an ESG-PPP screening matrix for procuring a P3 model, based on evaluated ESG project aspects, is illustrated in Figure 18 below. If the project objective indicates significant environmental, social, or governance impacts, using the ESG screening matrix for P3 procurement is necessary. If the decision indicates no ESG impacts, the project team may prioritize other procurement options without using an ESG screening matrix. Therefore, the decision chart ensures that if any of the questions result in a "Yes" answer, the assessment must be conducted. If all questions result in a "No" answer, then the assessment is not required.



Figure 18. Decision Flow Chart for use of ESG Screening Matrix in P3 Model Procurement

Quali	itative Criteria	Objectives to be achieved		
	Environmental Sustainability	The goal is to select a project delivery model whose organizational		
1	Environmental Criterion	structure and sequencing off activities aligns with the		
	Source: Canada green funding programs and Identified	environmental goals of a social infrastructure project.		
	challenges in P3 social infrastructure projects			
<ul> <li>1.1 Green adoption incentives <ul> <li>1. Green expertise partnership [113] [114] [91]</li> <li>[52] [13] [30] [39]</li> </ul> </li> <li>2. Green funding [115] [50] [48] [116]</li> </ul>		These criteria will evaluate the extent to which project delivery models prioritize environmental sustainability in infrastructure projects. By comparing the sequencing of activities and organizational structures across different project delivery models, the goal is to identify which models are more effective in promoting environmental sustainability through partnerships with green experts and access to green funding sources. This assessment aims to inform decision-makers about the strengths and weaknesses of each model in integrating green practices and fostering environmentally friendly infrastructure development.		

Table 12 Qualitative ESG Criteria for Canadian Social Infrastructure Projects

DBB	DB	DBF	P3	Progressive P3	P3 Bundle
The owner may	The contractor	Green expertise	Innovative private	Green expertise	Green expertise
engage green experts	may have more	partnerships and	sector green funding	partnerships roles	partnerships and
independently for	influence in	funding	investment may be	and risks may be	funding
design and	selecting green	mechanisms may	employed and may	established	mechanisms may
construction phases.	partners and	be established by	collaborate broadly	collaboratively	be established for
The owner typically	accessing	the financing entity.	with private	during the planning	the bundled
bears the risks	funding for	The entity providing	environmental	phase. Early	projects collectively.
associated with green	sustainable	financing may	organizations to	involvement of	
technologies and may	solutions.	influence decision-	provide expertise and	green expertise	
allocate funding		making regarding	guidance on	allows for greater	
accordingly.		green practices.	implementing	input in decision-	
			sustainable practices.	making regarding	
				green initiatives	
				from the beginning.	

2 Social Sustainability 2 Social Criteria Source: Community Benefits Agreement and Canada			The goal is to select a project delivery model organizational structure and sequencing off activities that aligns with the social goals of a social infrastructure project		
Infrastructure p	rogram plan [1] [63]	[43] [94] [51]	5		
<ul> <li>2.1 Community benefits</li> <li>1. Long/short-Term CBA</li> <li>2. Collaborative/Non-collaborative CBA</li> </ul>		This criterion evaluates how project delivery models achieve community benefits based on their structures, processes, and stakeholder engagement. Factors considered include community engagement levels, inclusivity in decision-making, and past performance in delivering community benefits. Models that actively promote more opportunities for marginalized groups, such as indigenous communities, contribute more to community benefits.			
DBB	DB Oiseten sist	DBF	P3	Progressive P3	P3 Bundle
DBBDBDBFLimited involvement of the community during the design and construction phases. Opportunities for local hiring may depend on contractors' practicesSingle-point responsibility streamlines decision- making, enabling faster responses to community input addressing of concerns, reducing bureaucratic delays.Financing considerations may impact the level of emphasis on community benefits; economic viability often takes precedence.		The long-term integration of project phases may have long-term community job opportunities. Leading to training and long-term local hiring for non-core P3 services.	Early collaboration in progressive P3 models allows negotiation of community benefits during procurement, incentivizing the private sector to deliver social benefits as part of the agreement.	Opportunities for community engagement are maximized within the scope of the bundled small projects.	

3. Responsible		The goal is to select a project delivery model organizational						
Governance Criteria		structure and sequencing off activities that aligns with the						
	Source	: Identified challenges	in P3 social infrastructure	transparency goals of a	a social infrastructure p	roject.		
	project	-						
31	Stakeh	older engagement: T	ransnarency and	This criterian assasses stakeholder angagement in the project				
0.1	Accour	ntability [16] [49] [33]	ranoparonoy and	delivery model aiming	delivery model aiming to choose a model that achieves			
	1	Farly Engagement		transparency involves	stakeholders in key de	cisions and		
	2	Communication Char	nnels	maintains accountabilit	v			
	3	Decision-Making Stru			y.			
	4	l ong-Term Stakehold	der Management					
DBB		DB	DBF	P3	Progressive P3	P3 Bundle		
DBB's	;	DB offers more	Communication in DBF	P3 projects	Private sectors	Stakeholder		
seque	ntial	collaboration	may need to address	necessitate effective	collaborate in	engagement		
nature	e limits	opportunities.	both construction	long-term relationship	shaping the project	approaches may		
transp	arency,	Transparency is	progress and financial	management with	agreement after the	vary based on the		
accou	ntability	typically enhanced	aspects, potentially	stakeholders.	RFQ stage, with	characteristics of		
as the	;	in DB due to its	increasing the	Maintaining positive	ongoing	the individual		
phase	s is	integrated nature,	complexity of	relationships and	stakeholder	bundled projects		
segme	ented.	and accountability	stakeholder	addressing concerns	engagement and			
		is shared within the	engagement.	over an extended	joint development of			
		design-build team,	Transparency in	project lifecycle	project aspects.			
		promoting collective	financial arrangements					
		responsibility for	becomes a critical					
		project outcomes.	aspect.					

## 2.5. Summary of identified gaps in the literature

Several studies have examined the positive impact of the ESG integration of sustainable practices into project management processes [100] [55]. However, research indicates that these ESG sustainability considerations have not been fully integrated into the procurement and governance practices of P3 infrastructure projects [107]. There is a recognized need for a more comprehensive approach to integrating ESG considerations into infrastructure projects [117]. P3s have been explored as a potential tool to address ESG issues, with suggestions for further research into how they can effectively manage ESG challenges across the project lifecycle, from planning to implementation [55] [118] [104]. Likewise, in the Canadian context, three main gaps have been identified in existing procurement assessments: the absence of Canadian-specific ESG procurement evaluative criteria for P3 social infrastructure projects, the lack of integration of ESG aspects in the procurement process, and the exclusion of ESG aspects in the multi-criteria analysis for selecting the best P3 model to compare against VfM considerations. This thesis aims to contribute to the integration of ESG sustainability considerations in procuring P3 social infrastructure projects in Canada by evaluating how well the P3 model achieves ESG sustainability in infrastructure projects. Figures 19 and 20 below summarize identified gaps, proposed contributions, and the current necessity for integrating ESG factors in the procurement evaluation of a P3 model. These contributions address literature gaps in three key areas: establishing environmental, social, and governance criteria for assessing responsible sustainability management in Canadian P3s; introducing a new assessment process, a screening matrix for evaluating ESG criteria in social infrastructure projects; and developing an AHP selection model tool for selecting P3 model types, considering both ESG qualitative and VfM quantitative aspects.



Figure 19. Literature review gaps



Figure 20. Need for Integrating ESG Factors in Procurement evaluation of a P3 model

# Chapter 3: Methodology

# 3.1. Introduction

This chapter outlines the methodology used to achieve the second and third objectives of this thesis, divided into two sections for clarity. The first section, 3.2, details the development of the ESG-PPP screening matrix. It outlines the process of identifying the ESG criteria, the weighting and scoring mechanism, and the decision thresholds. Additionally, it explains the rationale behind the ESG-PPP assessment matrix structure. Section 3.3 will cover the selection process of the AHP methodology and explain why it was chosen. It will detail the steps involved in its selection process, highlighting the importance of low inconsistency performance in AHP scales. The AHP scales will therefore be described, including the criteria used for the selection of the comparative scale that will be examined in the fourth chapter. Also, the hierarchy structure for selecting sustainable and responsible P3 models will be illustrated and explained.

## 3.2. Development of the Screening Matrix

The screening matrix is developed to evaluate the potential of P3s to achieve sustainability and responsible project management in Canadian social infrastructure projects. It will use three dimensions: Environmental, Social, and Governance, each with its own set of evaluative criteria. These evaluative criteria may vary depending on the different types of infrastructure project evaluated. Thus, for the scope of this study, the ESG criteria used are defined based on common ESG objectives and project requirements identified for hospitals, schools, public transit, social housing, recreation, and cultural centers. Each criterion will have a set of questions, each with response indicators rated on a 1-5 scale. The total score for each ESG dimension total will be calculated by normalizing and combining the criterion responses. Then, each dimension total will be combined to determine if a P3 model is satisfactory to deliver sustainable and responsible project management. This assessment approach will provide a structured framework for decisionmaking, enabling stakeholders to make informed choices based on an analysis of the nonfinancial ESG aspects that play an important role in improving overall project success and value creation. The following steps were followed in the development of the ESG-PPP screening matrix as illustrated in the Figure 21 below.



Figure 21. Process Flow Diagram for Screening Matrix Development

**Step 1:** The criteria identification process systematically gathered and refined relevant ESG criteria for the assessment matrix. First, a thorough literature review was done to extract the ESG criteria. This process included analyzing federal sustainability reports, provincial and federal infrastructure plans, sustainability publications from the Canadian infrastructure industry, and relevant guidelines. Additionally, project-specific ESG objectives, requirements, and current P3 news coverage from sources like CCPPP and other news websites were considered. Throughout the process, careful attention was given to refining the identified criteria, eliminating duplicates,

consolidating similar ones, and ensuring alignment with the research goals. Although stakeholders were not directly involved, the identified criteria were checked for relevance and coherence to ensure they collectively represented a complete ESG set of factors aligned with the research objectives. The following were the identified ESG evaluative criteria scope for the Canadian context, as illustrated in Figure 22 below:

- 1. The environmental dimension focus on green practices with positive ecological, climatic, and resilience impacts assessing green adoption incentives including green funding and green private expertise. The questions measure the availability and degree to which a P3 model allows for partnership with private entities with environmental expertise such as CAGBC and the incentives provided within the P3 model organization structure to have private funding with environmental conscious investors to encourage sustainable and environmentally friendly practices.
- 2. Social dimensions emphasize community benefits, assessing the adoption of community benefit agreements, community engagement, and inclusivity. The questions measure how the P3 structure and sequencing of activities influences the nature and scope of using the community benefits agreements, either emphasizing and prioritizing immediate gains or long-term sustainability and if they are developed collaboratively with community stakeholders.
- 3. Governance dimensions assess ethical practices and adherence to governance frameworks, ensuring transparency and accountability in stakeholder management and decision-making processes. The questions measure how the P3 model organisation structure influences the overall decision-making processes within the project team and other stakeholders, involving the formulation, discussion, communication, finalization, and implementation of project agreements.



Figure 22. ESG dimension and evaluative criteria for the ESG-PPP screening matrix The criteria within each of the ESG dimensions align with the broader goals of the Canada Infrastructure Plan. However, as mentioned earlier, these ESG evaluation criteria may change. The project team has the flexibility to suggest new ESG criteria to tackle current issues and challenges specific to a project category. This is important as the P3 model evolves and adjusts over time.

**Step 2:** Weighting involves assigning relative importance to each criterion. Weighting is applied prior to scoring and determines the impact that each criterion will have on the final decision. Criteria with higher weights contribute more to the overall assessment, while criteria with lower weights have less influence. Hence, assigning weights to criterion is a subjective process and depends on the decision problem [119]. Existing assessment matrices like P3 Canada's and British Columbia's use both quantitative and qualitative subjective weighting methods [69] [66]. The objective method uses a mathematical function to calculate weights based on criteria information [120], while the subjective method relies on decision makers' numerical weighting preferences [121]. Therefore, different contexts may call for different approaches based on available data; however, the selected method should align with the overall decision goals [122]. In this ESG-PPP assessment matrix, all identified project objectives and challenges are considered equally important. Therefore, the preferred weighting preference assumes that all ESG criteria hold equal importance for evaluating the sustainability and responsibility of the P3 model in social

infrastructure projects. Thus, a quantitative objective weighting method, specifically the equal weight approach [120], is chosen. This method is often used when criteria are considered equally important, allowing stakeholders to assign equal weights to each criterion. By using this approach, the subjective nature of the matrix is reduced compared to using subjective weighting methods. In different ESG-PPP screening process, other weighting methods can also be considered as ESG decision considerations evolve. Table 13 below summarizes the available weighting methods and their application in the existing Canadian P3 suitability matrix.

Table	13.	Criteria	Weighting	Methods
-------	-----	----------	-----------	---------

Weighting method	Description / Equation	Assumptions	Туре	Application of weighting methods in the existing Canadian P3 suitability matrix
Points	A total of 100 points is	The importance	Numerical	P3 Canada P3
allocation	distributed among the	of criteria in P3		suitability
	criteria based on their	selection varies	Subjective	assessment
	relative importance.			matrix
Ordinal Scale	Criteria are weighted	The importance of criteria in P3	Qualitative	British Columbia
obalo	categories.	selection varies	Subjective	procurement
				analysis.
Equal weights	$W_j = \frac{1}{n}$ , n is the number of criteria [120]	All criteria equally impact P3	Quantitative	Proposed ESG- PPP screening
		selection.	Objective	matrix

**Step 3:** Scoring involves assigning numerical values to criterion based on their performance. It is used to quantify the degree to which each criterion meets the predefined ESG objectives. Different scores can be applied to different criterion [119]. However, this matrix uses the same scoring rating for all criteria similar to the existing P3 Canada choice of scoring scale. The scoring scale is a 5-point scale to rank responses numerically, indicating how effectively the objectives of an ESG criterion will be met. Each number from 1 to 5 represents a level of response or opinion, with 1 typically indicating the lowest or least favorable impact, and 5 indicating the highest or most favorable impact. The scoring scale allows stakeholders to express how the criterion will be

achieved based on the P3 model characteristics. The scores measure the extent to which the criterion is satisfied facilitating the quantification of the ESG impact for analysis to establish decision thresholds. Table 14 below shows the summarised criteria weights and the scoring scale considerations.

Criteria	Dimension weight	Scoring Subjective	Rationale
Environmental aspects: Green adoption incentives	33.3%	Scoring Range: 1 to 5 points (Weighted range: 0.33 to 1.665)	Incentives for adopting green practices significantly contribute to the overall environmental sustainability of the project, including access to green funding and expertise.
Social aspects: Collaborative/ Long-term CBA	33.3%	Scoring Range: 1 to 5 points (Weighted: 0.33 to 1.665)	The impact of community benefits varies across different phases of a project, and the collaborative efforts involved in achieving these benefits play an important role in promoting social sustainability.
Governance aspects: Stakeholder management	33.3%	Scoring Range: 1 to 5 points (Weighted: 0.33 to 1.665)	Effective stakeholder management is important for project governance success. When stakeholders are actively engaged in a transparent manner, they are more likely to hold project managers and key stakeholders accountable for achieving objectives and delivering results throughout the project lifecycle.

Table 14	<ol> <li>Criteria</li> </ol>	Weighting	and	Scoring

**Step 4:** Normalizing criterion scores is important when different scoring methods are used for various response indicators of the criterion. In this assessment matrix, the same scoring rating, with a maximum of 5 points, was applied uniformly across all indicators. Therefore, normalization of the scoring was not necessary.

**Step 5:** To establish decision thresholds, each of the six criteria contributes equally, contributing 16.7% to the overall evaluation. Scores of 4-5 contribute positively, scores of 3 contribute neutrally, and scores of 1-2 contribute negatively. Getting the total score begins by calculating the weighted

scores this includes multiplying the criterion scores by their corresponding weights. The criterion weighted score has the lowest possible score of 1, resulting in a weighted score of 0.167, and the highest possible score of 5, resulting in a weighted score of 0.835 for the criterion. Subsequently, the weighted scores are summed up to obtain the total weighted scores for each dimension. Which are then summed up to get an overall weighted score. The overall weighted score has the lowest possible overall score of 1.002, and the highest is 5.01. Therefore, thresholds in Table 15 below categorize P3 suitability for sustainability and responsible outcome based on scores ranging from 1.002 to 5.01. Scores below 2 indicate unsatisfactory P3 performance, while scores above 4 indicate satisfactory P3 performance. These thresholds aid in screening P3 model sustainability, ensuring alignment with environmental, social, and governance criteria.

Table 15. P3	Procurement	Decision	Thresholds for	Responsible	Sustainability
--------------	-------------	----------	----------------	-------------	----------------

Decision thresholds	Description
4-5	Satisfactory
3	Adequate: Represents a moderate level of performance where
	certain aspects of the project may require additional efforts or
	resources to bring the P3 model up to satisfactory levels.
1-2	Unsatisfactory

This screening matrix developed was informed by P3 Canada's established screening assessment matrix, which is considered an industry standard [69]. Although involving stakeholders in developing the matrix structure components would benefit the matrix development [20], it was not implemented in this specific thesis due to constraints such as time, resources, and the unavailability of key stakeholders. Conducting extensive stakeholder discussions was not practical. Instead, the available P3 Canada matrix reference provides a familiar and credible structure for the ESG-PPP assessment matrix, ensuring consistency, acceptance, and alignment with industry practices, especially for stakeholders already accustomed to its structure.

After following the outlined steps and developing the screening matrix for assessing P3 suitability in terms of sustainability and responsibility, the next objective is to create a model for selecting the appropriate P3 delivery model. This model will consider both qualitative and quantitative VfM analysis results, facilitating the procurement decision multi-criteria options analysis outlined in the P3 procurement guide [68] [123].

### 3.3. Analytical Hierarchy Process for procurement of P3 options

Various multi-criteria decision-making methods are available, each suited for different decision problems [124]. In the case of the P3 model type decision problem, the AHP method was chosen. This decision was made because the problem involves a set of criteria and alternatives that can be organized in a hierarchy structure. Within this structure, the relative importance of each criterion and alternative to each criterion can be calculated using pairwise comparison. The resulting number of pairwise comparisons is relatively manageable, given the number of criteria and alternatives [125], which maintains ease of comparison and prevents complexity. Specifically, this selection problem involves evaluating five P3 model alternatives across four criteria: environmental, social, governance, and value-for-money aspects. These criteria and alternatives can be organized into a hierarchical structure, where consistent judgments and comparisons are made at each level. The AHP methodology excels in managing decision problems that include both quantitative and qualitative criteria within a hierarchical structure. It effectively handles subjective opinions from stakeholders and prioritizes criteria or alternatives based on their relative importance. Through this structured approach, stakeholders' subjective opinions on criteria importance are quantified into numerical weights for objective selection, making it more inclusive and better suited to real-world complexities [126] [127]. Given that the decision problem of the P3 model aligns well with the characteristics offered by AHP, it is chosen as the preferred approach. The key process steps followed in the development of the AHP-PPP selection model is illustrated in Figure 23 below.



Figure 23. AHP Development Process Flow Diagram

With reference to the above stages. The following are the steps followed in creating the selection model for this P3 model type selection decision.

**Step 1**: **Hierarchy Structure:** The initial step involves the formulation of the selection problem within a hierarchical structure. The hierarchical structure proposed within the scope of this study consists of four levels as seen in Figure 24 below.

Level 1: The main objective of the selection problem is to choose a P3 model that considers the best sustainability and responsible project delivery practices for a social infrastructure project. This is put as the highest level.

Level 2: This level represents the main criteria or aspects that contribute to the overall decision of the main objective. The four main criteria for comparison are the identified qualitative environmental, social, and governance criteria, along with the quantitative aspect of the Value-for-money achieved from previous P3 projects. These criteria are independent to avoid redundancy as this can lead to inconsistencies and make the decision-making process less reliable. If criteria are highly correlated or redundant, it can lead to double-counting or overemphasizing certain aspects of the decision [128]. Each criterion is defined on how it is to be evaluated to fit within the context of this hierarchy structure as follows. It includes all considerations constituting the proposed National ESG category award by the CCPPP [129].

a. Environmental

The environmental criteria assess each P3 model's capacity to address Canadaspecific aspects such as environmental protection or enhancement, climate change mitigation, wildlife conservation, and natural infrastructure attributes [129]. For example, when working on a construction project in a sensitive environmental area. The choice of P3 model can influence how well the project team, phases, and sequencing of activities may favor protecting local wildlife habitats and minimize environmental impact. As such a model like DBFM may prioritize environmental protection by integrating sustainable construction practices and minimizing environmental impact throughout the project lifecycle. DBFOM models could emphasize long-term environmental management through ongoing operations and maintenance practices. Progressive DBFM and Progressive DBFOM models might
offer innovative solutions for addressing specific environmental challenges, while P3 Bundle models may provide comprehensive approaches to managing environmental considerations across multiple projects within the bundle. Each P3 model brings unique advantages and considerations for addressing environmental concerns, influencing project delivery strategies accordingly.

b. Social

The social criteria involve assessing each P3 model to address Canada specific aspects like Community engagement, social inclusion involving indigenous peoples, and other social impacts. For example, social infrastructure projects require to attain social value outcome governed under a community benefits agreement. A DBFM model often involves a single private entity responsible for designing, building, financing, and maintaining the infrastructure. In this setup, the private partner may have more control over subcontracting and procurement decisions, potentially impacting job training opportunities and the involvement of local businesses and social enterprises during construction. On the other hand, a DBFOM model, where the private partner also operates and maintains the infrastructure, may provide additional incentives for the private entity to allows for ongoing collaboration between the public and private sectors throughout the project lifecycle, facilitating the long-term non-core services contracts promoting more operational opportunities. Progressive P3 models enable stakeholders to iteratively refine social value deliverables and ensure that they remain relevant throughout the project. On the other hand, P3 bundle models can enhance the coordination of social value considerations across different sub-projects and maximize their collective impact on job creation, local procurement, and community development.

63

### c. Governance

The governance criteria involve assessing each P3 model's ability to address Canada specific P3 challenges focusing on stakeholder management to ensure transparency and accountability. In DBFM projects, while governance aspects may be outlined in the contract, the direct accountability on governance practices is limited during construction. The DBFOM models that include operation and maintenance responsibilities can enhance governance by promoting ongoing accountability and transparency throughout the project lifecycle. The progressive P3 models can facilitate the incorporation of robust governance mechanisms to prevent fraud and corruption as all stakeholders are involved in the early stages to ensure transparency and accountability before the project agreement is finalised. The P3 bundle may have a comprehensive governance framework that spans across all bundled projects, promoting consistency, efficiency, and accountability in governance practices.

Level 3: This level represents the available options that are to be selected from. It consists of the commonly used P3 model options for delivering social infrastructure projects in Canada, which are DBFM, DBFMO, progressive P3 models (progressive DBFM and progressive DBFMO), along with the P3 bundle model.



Figure 24 Hierarchical Structure for Selecting Sustainable and Responsible P3 Models

The hierarchical structure is an important part of our AHP selection model, significantly influencing the quality of decision-making. The number of pairwise comparisons made in an AHP is determined by the number of options or criteria on the hierarchy structure. Ongoing research also highlights concerns about potential inconsistencies possible in the AHP, particularly when dealing with numerous pairwise comparisons due to a large number of criteria in a hierarchy structure [130]. To avoid complexity and mutual dependency, this thesis limits the evaluative to four mutually independent criteria, without any sub-criteria. Hence, after all levels are defined, the decision-making process to choose the best P3 model aligning with responsible and sustainable practices for social infrastructure development in Canada follows the steps outlined in Step 2. Figure 25 below illustrates the key stages following to the creation of the hierarchy structure for the selection problem.



Figure 25 Key Stages Following Hierarchy Structure Establishment

Step 2: Establish a Scale: The calculation of weights depends on the chosen scale, which in turn influences the dispersion and uncertainty of the priority weights [131]. Hence, the selection of the appropriate scale is an important aspect of the AHP methodology and can impact the accuracy and reliability of the decision-making process. This step requires to identify a numerical scale to express the relative importance or preference between the elements in order to convert subjective judgments into quantitative scores. Several types of scales have been employed to represent the relative significance of pairwise comparisons of criteria in an AHP process [132]. Figure 26 categorizes these scales based on the highest value assigned to the decision matrix.



### Figure 26. Different types of AHP scales

Among the scales falling under category B, the generalized balanced scale improves both weight dispersion and uncertainty compared to the fundamental AHP scale. Weight dispersion measures the evenness of weight distribution across criteria, while weight uncertainties reflect the confidence in the assigned weights. Higher weight uncertainties imply greater ambiguity, while lower uncertainties indicate more confidence. Similarly, higher weight dispersion suggests wider variation in assigned weights, whereas lower dispersion implies more even distribution. Practical AHP applications demonstrate that the generalized balanced scale leads to better consistency ratios, indicating improved reliability in decision-making processes [133] [131].

Hence, this study will compare P3 model type results using both the fundamental AHP scale and the generalized balanced scale to assess if improved consistency ratios influence the results differently. The fundamental scale where 1 means both criteria are equally important or have equal preference, and 9 indicates one criterion is extremely more important or preferred than the other. Intermediate values represent varied degrees of importance or preference [134]. The balanced scale, proposed by Salo and Hämäläinen, addresses the issue of disproportionate weight changes in the Fundamental AHP scale. They found that small adjustments in judgments at the extreme ends of the scale lead to significantly larger changes in weights [135]. To achieve

equal weight distribution, it is recommended to use a point scale ranging from 1 to 9, structured according to integer ratios provided in Table 16 below, under the Balanced scale preference input. This adjustment addresses the uneven weight distribution issue identified in the Fundamental AHP scale for problems with only two criteria. However, when a decision problem involves more than two criteria, the weight distribution becomes uneven [133]. The generalized balanced scale builds upon this concept, extending its application to decision problems with multiple criteria while maintaining consistent weight distribution by accounting for the number of criteria involved [135]. This thesis plans to use case studies and compare results to assess whether the two scales produce different outcomes given the identified weight distribution variations. Existing research has focused relatively more on studying inconsistency in pairwise comparisons, particularly those aimed at reducing the iterative processes involved [130] However, there is a lack of studies exploring inconsistency results when using different AHP scales across various case studies [136]. Table 16 below shows the mathematical function used to change the range scale of 1 to 9 into integer ratios for the balanced scale and the generalised balanced scale.

х	Definition	Fundamental AHP Scale $C = x$	Balanced $\left(C = \frac{9+x}{11-x}\right)$	Generalised balanced (bal- $\left(\begin{array}{c} C = \frac{9 + (n - 1) x}{9 + n - x} \end{array}\right)$ Where level 2, n=4 Level 3, n=5	
			n = 2	n = 4	n = 5
1	Equal	1	1	1	1
2	Equal/Moderate	2	1.22	1.36	1.42
3	Moderate	3	1.50	1.8	1.91
4	Moderate/Strong	4	1.86	2.33	2.3
5	Strong	5	2.33	3	3.22
6	Strong/Very Strong	6	3.00	3.86	4.125
7	Very Strong	7	4.00	5	5.29
8	Very	8	5.67	6.6	6.83
	Strong/Extreme				
9	Extreme	9	9	9	9

Table 16. The comparative scales

**Step 3: Pairwise matrix:** This step compares each criterion against every other criterion using the established scale in a matrix form. A pairwise comparison answers the question of how essential one criterion, say value-for-money for a DBFM, is in relation to the green adoption incentive that DBFM model will has given its phase integration characteristic.

- The pairwise comparison begins with paired comparison matrices for each hierarchy level as illustrated in Figure 27 below were the rows and columns represent the comparison between two criterias. For this study a total of 6 paired comparison should be done for level 2 and 4 for level 3 with respect to level 2. A higher number is assigned if one of the criteria is more important than the other and the reciprocal is assigned if the opposite is considered more important.
- The comparisons on level 2 include: "Green practices" vs. "Community benefits", "Green practices" vs. "Value-for-money", "Green practices" vs. stakeholder engagement, "Community benefits" vs. "Value-for-money", Community benefits vs. Value-for-money, stakeholder engagement vs. "Value-for-money" in relation to the primary objective of selecting the optimal sustainably responsible P3 model. The comparison on level 3 should be compared to the level 2 criteria in a similar manner.

The effectiveness of this approach will highly depend on the responses to the criterion comparison questions, which are formulated on the chosen scale of judgment.

Criterion "n"	$A_1$	$A_2$	A3	$A_4$
A1 (	<i>S</i> <sub>1</sub>	$S_5$	S9	S <sub>13</sub>
$A_2$	$S_2$	$S_6$	S <sub>10</sub>	$S_{14}$
$A_3$	$S_3$	$S_7$	$S_{11}$	$S_{15}$
A4	$S_4$	S <sub>8</sub>	$S_{12}$	S <sub>16</sub>
· · · · · · · · · · · · · · · · · · ·	_			_

Figure 27. Pairwise matrix

**Step 4: Normalisation process:** This step follows to standardise the preference values in the pairwise matrix in a range of 0 to 1, known as eigenvectors. These decomposed matrices as seen in Figure 28 below accurately reflect the relative importance of the compared options with respect to the comparison goal on each level of the hierarchy structure. The formula for calculating the weights as proposed by Saaty [134] is expressed below as:

Normalised weight, 
$$n = \frac{Preference Score}{Sum of scores for the column}$$
 (1)

	$A_1$	$A_2$	$A_3$	$A_4$
$A_1$	( W1	$W_5$	$W_9$	W <sub>13</sub>
$A_2$	$W_2$	$W_6$	W <sub>10</sub>	$W_{14}$
$A_3$	$W_3$	$W_7$	$W_{11}$	W <sub>15</sub>
$A_4$	$W_4$	$W_8$	$W_{12}$	W16

Figure 28 Normalised Matrix

After, the principal eigenvector is calculated, the normalised principal eigenvector [134] is calculated as expressed below as:

Normalised weight, 
$$n = \frac{1}{3} \times Normalised$$
 weights matrix (2)

Once the weights are known for each level, the eigenvectors are employed in the determination of the major eigenvalue [128], which serve an important input in checking the coherence of subjective judgments made during pairwise comparisons. The following formula [134] summarised the mathematical procedure used.

$$\lambda_{max} = \sum (eigen \ vector \ \times \ sum \ of \ scores \ of \ the \ column)$$
(3)

**Step 5: Consistency Verification:** This step is taken to ensure logical consistency between the preference values of the decision-makers in the normalized matrix and the derived eigenvectors [128]. The measure of consistency is given as a consistency index as shown in formula below as proposed by Saaty [134], and used to calculate the consistency ratio [134]. Some inconsistency is expected and allowed in AHP analysis. However, if the consistent ratio is lower or equal to 10%, then the inconsistency is acceptable. The random index Table 17 below highlights the values used.

$$CI = \frac{\lambda_{max} - n}{n - 1} \tag{4}$$

$$CR = \frac{CI}{RI} \tag{5}$$

Table 17. Consistency indices for a randomly generated matrix [134]

n	1	2	3	4	5	6	7	8	9	10
RI	0	0	0.58	0.9	1.12	1.24	1.32	1.41	1.45	1.49

Step 6: Combining priority vectors and choosing the highest priority: The priority vectors (weights) obtained for criteria and alternatives at each level are combined. This involves aggregating the weights assigned to different criteria and alternatives within the hierarchy. The combined set of weights represents the overall priorities of the alternatives in the decision hierarchy. Finally, the alternative with the highest overall priority is considered the most favorable or suitable choice.

Steps 1-6 of the AHP method explain the key processes involved, along with why they're important. AHP provides a clear way to measure subjective opinions, making it ideal for choosing a P3 model for sustainable social infrastructure projects. Considering this, and the need for a simple decision-making tool, AHP was chosen. Additionally, this iterative process, which involves

creating a priority list at each level of decision-making and selecting the best alternative, has been automated using an Excel tool developed alongside this thesis (see appendix III). This tool will be used in the next chapter to analyze selected case studies, using the two scales as discussed above. The aim is to analyze and discuss the outcomes of the selected P3 model for the social infrastructure case studies.

# Chapter 4: Application of AHP-PPP Excel Tool

### 4.1. Introduction

This chapter aims to demonstrate the functionality and reliability of the developed AHP-PPP Excel tool for selecting a P3 model with the best sustainable outcomes in delivering and managing social infrastructure projects. Section 4.2 will outline the criteria used in selecting the case studies, each case is described together with the scope and objectives and present the outcomes of the P3 model selected using the developed AHP-PPP excel tool made using the Macros and VBA Scripts, which automated the AHP process tasks. The focus of the results will be on the AHP scales consistency in assessing the ESG and VfM project objectives across the selected three case studies. Following that, Section 4.3 will discuss the results comparison of the two AHP scale, as discussed in Chapter Three.

### 4.2. Case Study Selection

Three social infrastructure projects have been chosen as case studies for analysis within the proposed AHP-PPP model selection tool. The selection criteria include the scale and type of the social projects, availability of P3 model value-for-money data, and the ESG governance mechanism used. Based on these criteria, the selected case studies are the South Niagara Hospital, Finch West Light Rail Transit, and the Saskatchewan Joint Use School Projects.

## 4.2.1. Case study 1: South Niagara Hospital

### 4.2.1.1. Case description

The South Niagara hospital is located in Ontario with a size of 1.3 million square feet [137]. It is currently under construction after a 10-year planning period [138] and is scheduled for completion by 2028. It represents an important example of the implementation of a 35-year DBFM P3 model in the healthcare sector. The project aims to meet the growing healthcare demands of the South Niagara region and has been

74

nominated for the ESG National Award category at the 2023 National P3 Awards [139]. This recognition highlights the project's innovative approach, sustainable design considerations, and positive community impact, including collaboration with indigenous stakeholders in project design [138]. The project team consists of Niagara Health, overseeing development and operations, Infrastructure Ontario managing project management, financing, and procurement, and EllisDon Corporation as the builder with Parkin Architects Ltd. and Adamson Associates Architects as the project architects and EllisDon Facilities Services as the service contractor [140].

#### 4.2.1.2. Scope and Objectives of Niagara Health

The South Niagara Hospital project aimed to achieve three key objectives. Firstly, it aims to employ innovative designs and technology to prioritize sustainability and minimize environmental impact by efficiently using resources and adopting green building practices. The goal is to meet the LEED Silver standards set by the Canada Green Building Council and obtain WELL v2 certification [137] [141]. Secondly, it aims to involve the local community in the planning process to ensure the hospital meets their needs. Thirdly, the delivery of the project requires to collaborate with various stakeholders, including government bodies such as the ministry of health, community members, hospital staff, and partners, to ensure project success. Likewise, by opting for a DBFM model, the project's value-for-money objective is to achieve 15% to 22% savings [141]. Table 18 below summarises the key objectives and the subsequent evaluative consideration used as reference during the subjective preference judgement of ESG weights.

Objectives	ESC evolutive consideration for South Niczara project
Objectives	ESG evaluative consideration for South Magara project
Sustainability and environmental impact reduction	Environment: Comparing the extent to which the one P3 model encourages the private partner to invest in innovative technologies and sustainable practices that improve long-term performance and reduce environmental impact.
Community involvement in planning	Social: Community engagement and feedback mechanisms: Comparing the extent to which one P3 model facilitates meaningful community involvement in the planning, design, and decision-making processes related to the hospital project against the other. Social: Local benefit provision: Comparing how one P3 model ensures to deliver relevant community opportunities.
Stakeholder collaboration	Governance: Effective communication strategies: Comparing how one P3 model facilitates collaboration and engagement among diverse stakeholders including the hospital staff. Governance: Comparing how one P3 facilitate transparency and inclusivity of decision-making processes within the P3 framework considering early stakeholder engagement.
Financial savings	Value-for-money: Comparing the P3 model historical data on attaining value-for-money and long-term affordability including operational efficiency.

Table 18. Key Objectives and Proposed ESG Evaluation Criteria for the Hospital Project

4.2.1.2 Application of AHP-PPP Selection Model

The AHP-PPP model is applied to the South Niagara Hospital project to compare and rank the available P3 model alternatives; DBFM, DBFOM, Progressive P3s, P3 Bundle model, based on how well they align with ESG evaluative criteria and value-for-money aspects. The main goal of the South Niagara Hospital project is environmental sustainability, aiming to reduce environmental impact by using resources efficiently and adopting green building practices. This is evident in the project's efforts to achieve two environmental building standards: the LEED Silver standards and WELL v2, showing a strong focus on environmental responsibility. While other objectives are important, they may not be as central as environmental sustainability. The WELL building standard version 2 is a recent building design standard that focus on design and operational considerations that focus on human health and well-being [142]. Table 19 below shows the preference scale used as input = c, in the AHP-PPP excel tool with reference to the Saaty Fundamental scale and the generalised balanced n scale.

Table 19. Verbal Preference to Numerical Conversion Input (c) used in AHP-PPP Excel Tool

x	Verbal preference	Fundamental	Generalised balanced (balanced n	
		AHP Scale	$\left( C = \frac{9 + (n-1)x}{9 + n - x} \right)$	
		C = x	Where level 2, n=4	
			Level 3, n=5	
		_	n = 4	n = 5
1	Equal	1	1	1
2	Equal/Moderate	2	1.36	1.42
3	Moderate	3	1.8	1.91
4	Moderate/Strong	4	2.33	2.3
5	Strong	5	3	3.22
6	Strong/Very Strong	6	3.86	4.125
7	Very Strong	7	5	5.29
8	Very Strong/Extreme	8	6.6	6.83
9	Extreme	9	9	9

Tables 20 and 21 show the priority values of the criteria and the summarized alternative weights, respectively, comparing the results from the Fundamental AHP Scale and the Balancedn scale.

Table 20. South Niagara Hospital: Local criteria priorities

		Fundamental	Balanced-n Scale	Differences (-/+)
		AHP Scale	n = 4	
1.	Environmental	0.394 (39.4%)	0.432 (43.2%)	-0.038
2.	Social	0.241	0.208	0.033
3.	Governance	0.167	0.172	-0.005
4.	Value-for-money	0.197	0.188	0.009
	Consistency Ratio	0.02	0.006	0.014

Both scales prioritize the environmental criterion as the most important consideration aligned with the project's objective. Likewise, the goal is to ensure that the decision support AHP-

PPP excel tool produce consistent and reliable results with low inconsistency value of less than 0.1, as indicated by the results the inconsistency value is less than 0.1. The difference between the two scales is also very minimal suggesting not much difference for results of the P3 model ranking considering a selection problem having four criteria. The priority values for the alternatives concerning each criterion are summarized in Table 22 and depicted in Figure 29 below

Fundamental AHP	Environmental	Social	Governance	Value-for-
Scale				money
DBFM	0.139	0.120	0.129	0.146
DBFMO	0.164	0.164	0.169	0.146
PR. DBFM	0.214	0.238	0.220	0.223
PR. DBFOM	0.289	0.286	0.289	0.291
P3 bundle	0.193	0.192	0.193	0.194
Consistency ratio	0.049	0.06	0.05	0.033
Balanced n Scale	Environmental	Social	Governance	Value-for-
n = 5				money
DBFM	0.137	0.137	0.129	0.142
DBFMO	0.180	0.180	0.190	0.206
PR. DBFM	0.203	0.231	0.217	0.183
PR. DBFOM	0.286	0.254	0.269	0.272
P3 bundle	0.195	0.197	0.195	0.197
Consistency ratio	0.029	0.019	0.033	0.018

Table 21. South Niagara Hospital: Local alternative priorities with respect to each criterion



Figure 29. South Niagara Hospital: Priority weights of P3 models and each criterion

The final results for both the Fundamental AHP Scale and the balanced -n scale results analysis show that the Progressive DBFOM is the most suitable P3 model for the project as seen in Table 22 below. The project is being delivered using a DBFM model.

P3 Alternative	Fundamental AHP scale	Balanced-n scale	Difference (+/-)
DBFM	0.130	0.137	-0.007
DBFMO	0.157	0.187	-0.03
PR. DBFM	0.216	0.207	0.009
PR. DBFOM	0.280	0.274	0.006
P3 bundle	0.187	0.196	-0.009

Table 22. South Niagara Hospital: Overall priority weights of P3 model option

## 4.2.2. Case study 2: Finch West Light Rail Transit

#### 4.2.2.1. Case Description

The Finch West LRT is a transit project situated in northwest Toronto, Ontario, spanning an 11-kilometre network with 18 stops. It includes the construction of key stations like the Humber College Station and Finch West Station, along with a 10,000 square foot storage and maintenance facility for the light rail vehicles [143]. Construction began in 2018 with an expected completion date in 2023. However, the project is still under construction and expected to complete in 2024 [144]. This serves as a significant case study highlighting schedule delays within a 35-year DBFM P3 model in the public transit sector. The project team includes the private sector entity, Mosaic Transit Group, chosen for their technical expertise, construction schedule, pricing, funding stability, and maintenance plans. Oversight during construction is provided by government agencies Metrolinx and Infrastructure Ontario on the public sector side. This project marks the second P3 social transit project to participate in Metrolinx's Community Benefits program, which is designed to offer economic opportunities, job training, and enhancements to the community [145].

#### 4.2.2.2 Scope and Objectives of Finch West Light Rail Transit

The objectives of the Finch West Light Rail Transit project are to improve public transit, minimize community disruptions during construction, ensure high-quality design within budget and schedule, and maintain the system long-term [145]. All these goals are required by the project agreement [146] to consider environmental considerations verified by third-party certification. However, there is no available information regarding the specific level of LEED certification achieved. The project opted for a DBFM model; therefore, the project's value-for-money objective is to achieve cost savings of up to 22.7%. However, due to the schedule delay of the DBFM model, the value-for-money may not have been fully met. On the other hand, participating in Metrolinx's Community Benefits program highlights the importance of "community benefits" in alignment with

broader project community goals. Table 23 below summarises the key objectives and the subsequent ESG evaluative consideration used as reference during the subjective preference judgement of ESG weights.

Objectives	ESG evaluative consideration for South Niagara project
Reduce traffic carbon emissions	Environment: Comparing how different P3 models will provide better innovation and incentive for transit infrastructure performance, operational efficiency, and sustainability.
Align with broader social and economic goals	Social: Comparing how different P3 models affect social factors like creating local jobs, promoting social inclusion, ensuring equity, and fostering community development given their contractual arrangements and phase integration.
Stakeholder management	Governance: Comparing how one P3 facilitate transparency and inclusivity of decision-making processes within the P3 framework considering early stakeholder engagement.
Financial savings	Value-for-money: Comparing on P3 model historical data on attaining cost savings taking consideration of the identified schedule delays and cost overruns in Transit projects.

Table 23. Key Objectives and Proposed ESG Evaluation Criteria for the Transit Project

4.2.2.3. Application of AHP-PPP Selection Model

In a similar fashion as in the earlier case study. The importance of judgment for criteria

with respect to the goal can be seen as a matrix as shown in the Table 24 below.

Table 24. Finch West Light Rail Transit: Local Criteria Priorities

	Fundamental	Balanced-n Scale	Differences (-/+)
	AHP Scale		
1. Environmental	0.351 (35.1%)	0.311 (31.1%)	0.04
2. Social	0.380 (38.0%)	0.329 (32.9%)	0.051
3. Governance	0.127 (12.7%)	0.172(16.5%)	-0.045
4. Value-for-money	0.142(14.2%)	0.187 (14.8%)	-0.045
Consistency Ratio	0.06 (6%)	0.003 (0.3%)	0.057

Both scales prioritize the social criterion and second the environmental criterion aligned with the ESG project's objective. The consistency ratios fall below the 0.1 threshold, indicating reliable consistency results. The priority values for the alternatives concerning each criterion are summarized in Table 25 and depicted in Figure 30 below.

Fundamental AHP	Environmental	Social	Governance	Value-for-money
Scale				
DBFM	0.129	0.138	0.153	0.147
DBFMO	0.169	0.169	0.226	0.147
PR. DBFM	0.220	0.278	0.197	0.254
PR. DBFOM	0.289	0.220	0.226	0.254
P3 bundle	0.193	0.195	0.197	0.197
Consistency Ratio	0.046	0.033	0.018	0.018
Balanced n Scale				
DBFM	0.161	0.163	0.174	0.172
DBFMO	0.185	0.184	0.214	0.172
PR. DBFM	0.212	0.243	0.199	0.228
PR. DBFOM	0.244	0.212	0.214	0.228
P3 bundle	0.198	0.198	0.199	0.199
Consistency Ratio	0.011	0.010	0.004	0.004

Table 25. Finch West Light Rail Transit: Local Alternative Priority with respect to Each Criterion



Figure 30. Finch West Light Rail Transit: Priority weights of P3 models and each criterion

The final results for the Fundamental AHP Scale and the balanced -n scale results analysis show that the PR. DBFOM is the most suitable P3 model for the project as seen in Table 26 below. The project is being delivered under a DBFM model.

Table 26. Finch West Light Rail Transit: Overall Priority weights of P3 model option	on
--	----

P3 Alternative	Fundamental AHP scale	Balanced-n scale	Differences
DBFM	0.138	0.166	-0.028
DBFMO	0.173	0.187	-0.014
PR. DBFM	0.244	0.223	0.021
PR. DBFOM	0.250	0.225	0.025
Alberta P3 model	0.195	0.198	-0.003

## 4.2.3. Case study 3: Saskatchewan Joint Use School Projects

### 4.2.3.1. Case Description

This project is the largest school project in Saskatchewan and was awarded the Silver Award for innovative Partnerships by CCPPP in 2015. It involves a 32-year DBFM agreement to construct 18 joint-use schools across 9 sites. Project 1 in Regina comprises three schools, while Project 2 covers four in Saskatoon, one in Warman, and one in Martensville. It is currently in its operation phase and was scheduled for completion in 2017. The value-for-money of the bundled DBFM was 13.5% of the total project cost, when compared to a conventional Design Build approach. The project team comprised of the Saskatchewan Ministry of Education as the public sector and Joint Use Mutual Partnership (JUMP) as the private sector [63].

### 4.2.3.2 Scope and Objectives of Saskatchewan Joint Use School Projects

The objectives of the Saskatchewan Joint Use School Projects are to deliver school buildings that provide conducive learning environments within budget and schedule. Likewise, managing a multi-site project with various stakeholders brings distinct challenges compared to a single-site project, requiring additional accountability and diligence for project agreement implementation [63]. Therefore, ensuring value-for-money and effective stakeholder management, including transparency and accountability, are the most important objectives given the nature of the project. Similarly, engaging with various school boards is important for project success, spanning from initial planning to execution, maintenance, and operation. On the other hand, although there's no formal commitment to a CBA, the projects observed engaging local businesses for economic benefits [63]. The environmental criterion was considered through the silver LEED certification that was to be achieved. The Table 27 below summarises the key objectives and the subsequent ESG evaluative consideration.

84

Table 27. Key objectives and ESG consideration for Saskatchewan Joint Use School Projects.

Objectives	ESG and VfM considerations
Multi-site project agreement implementation	Governance: Compares how a P3 model addresses the challenges of managing a multi-site project with diverse stakeholders, ensuring effective communication, oversight, and adherence to project objectives.
Financial savings	Value-for-money: Compares historical data on the P3 model capacity to deliver value-for-money over the project's lifecycle.
Community benefits	Social: Compares how a P3 model will engage with local communities for design inputs and feedback.
Building quality and comfort design	Environmental: Compares how different green design consideration will be engaged in different phases.

## 4.2.3.3. Application of AHP-PPP Selection Model

The priority values for each criterion's alternatives are summarized in Tables 28 and 29 and depicted in Figure 31 below. The final results for the Fundamental AHP Scale and the balanced -n scale results analysis show that the PR. DBFOM is the most suitable P3 model for the project as seen in Table 30 below. The project was delivered under a DBFM bundle model.

Table 28. Saskatchewan Joint Use School Projects: Local criteria priorities

		Fundamental	Balanced-n Scale	Differences (-/+)
		AHP Scale		
1.	Environmental	0.149	0.187	-0.038
2.	Social	0.233	0.225	0.008
3.	Governance	0.377	0.343	0.034
4.	Value-for-money	0.241	0.245	-0.004
	Consistency Ratio	0.046	0.022	0.024

Table 29. Saskatchewan Joint Use School Projects: Local alternative priorities with respect to each criterion.

Fundamental AHP	Environmental	Social	Governance	Value-for-money
Scale				
DBFM	0.104	0.128	0.089	0.082
DBFMO	0.141	0.171	0.133	0.137
PR. DBFM	0.230	0.252	0.295	0.137
PR. DBFOM	0.182	0.161	0.179	0.227
P3 bundle model	0.343	0.372	0.388	0.417
Consistency ratio	0.025	0.014	0.006	0.012
Balanced n Scale				
DBFM	0.141	0.160	0.132	0.122
DBFMO	0.170	0.160	0.182	0.165
PR. DBFM	0.222	0.209	0.193	0.165
PR. DBFOM	0.194	0.182	0.193	0.218
Bundle P3 model	0.273	0.290	0.299	0.331
Consistency ratio	0.007	0.004	0.002	0.005



Figure 31. Saskatchewan Joint Use School Projects: Priority weights of P3 models and each

criterion

P3 Alternative	Fundamental AHP	Balanced-n scale	Difference (-/+)
	Scale		
DBFM	0.099	0.138	-0.039
DBFMO	0.146	0.171	-0.025
PR. DBFM	0.184	0.195	-0.011
PR. DBFOM	0.187	0.197	-0.01
P3 bundle model	0.385	0.300	0.085

Table 30. Saskatchewan Joint Use School Projects: Overall priority weights of P3 model option

## 4.3. Analysis and Discussion

For Case 1, The South Niagara Hospital, the decision of the AHP-PPP model favored the Progressive DBFOM, as indicated in Table 22 above. The operational phase and the progressive procurement approach are expected to yield better outcomes in mitigating the environmental footprint and ensuring sustainable operations of the hospital, given the increased incentives for the private sector. This aligns with the key challenge and ESG objectives specific to hospitals identified in the literature, particularly focusing on improving operational efficiency. Additionally, in 2023, two hospital projects in Ontario were procured using the progressive DBFM model, indicating a preference for this model in delivering hospital project. However, the assessment of the progressive DBFM may have inadequately addressed ESG concerns and challenges.

For Case 2, the Finch west LRT, the decision of the AHP-PPP model favored the progressive DBFOM, as seen in Table 26 above. This decision aligns with the prevalent use of the progressive model for procuring transit-oriented projects in Ontario and the findings of the literature review that identified key challenges and objectives for transit projects, including the lack of consistent operational funding and the need for more reliable and sustained lifecycle funding, respectively. The operational phase of the DBFOM model will offer the public sector reliability in project funding from the private sector, potentially supporting the government during operation. Furthermore, the progressive procurement strategy aligns with the evidence-based decision-making objectives of

transit project implementation by ensuring early involvement of different stakeholders in project planning for more objective decision-making. However, the project was procured using the DBFM, with a primary focus on community benefits as the most critical criterion.

For Case 3, the Saskatchewan Joint Use School Projects, The AHP-PPP model's decision favored the P3 bundle model, as demonstrated in Table 30 above. This is consistent with the P3 bundle procurement approach adopted for this project, aimed at addressing the operational challenges outlined in Table 3. The progressive DBFM bundle emerges as the preferred choice. However, the progressive DBFOM bundle could also be considered, with the project team needing to involve school boards in the administrative operational team to ensure flexibility in managing the schools.

The historical data on the value-for-money of various Canadian P3 models, including DBFM, DBFMO, Progressive DBFM, Progressive DBFOM, and the P3 bundle model, along with the identified ESG project objectives documented in the case descriptions served as a useful guide for subjective preferences when making pairwise comparisons between these models for each case study. The results shows that the generalized balanced scale exhibits greater consistency and reliability when assessing P3 suitability for sustainable and responsible outcomes. Furthermore, the decision outcomes regarding P3 model selection remained consistent across all case studies for both scales, despite variations in criteria weight distribution, as observed by Goepel [135]. This indicates that both scales are dependable as long as decision makers refine their judgments to accurately reflect the relative importance of P3 alternatives. A sensitivity analysis, conducted using an AHP sensitivity analysis tool, revealed changes in rankings for the third and fourth alternatives maintaining the first rank for the first P3 model of the hospital and school projects. This indicates reliable decision-making unaffected by minor variations in criteria weights. As for the transit project the ranking of the first results changes with varying weight preferences (see appendix IV).

# Chapter 5: Conclusion and Recommendations

### 5.1. Conclusion

This thesis addressed the required shift in the screening of P3 social infrastructure projects in Canada. First, it identified existing qualitative P3 procurement evaluative criteria, which currently focus on six groups of criteria related to project characteristics, scope, cost, historical considerations, contract components, risk, and market capacity factors. While these criteria are effective for delivering some social infrastructure project objectives, there is a comprehensive need for addressing non-financial aspects that evaluate responsible practices and sustainability principles, to evaluate environmental and social impacts and align with Canada's commitment to delivering more sustainable and responsible infrastructure projects. Consequently, relevant ESG procurement objectives of social infrastructure projects were identified and defined to screen for a P3 model.

The specific ESG criteria identified for Canada cover various dimensions. Under environmental considerations, the criteria include green adoption incentives and life-cycle assessment. In addressing social aspects, the focus was on community benefits, including assessments of community impact phases and levels of community engagement. In terms of governance, stakeholder engagement criteria emphasized transparency and accountability. These criteria were then incorporated into the existing two-stage procurement process to evaluate the potential of P3 models in promoting sustainable and responsible practices within social infrastructure projects.

For the first stage of the procurement process, where a screening matrix is used to determine whether the project has the potential to be carried out as a P3 project. This thesis successfully developed a practical screening assessment form which is known as the P3-ESG suitability screening matrix. It will assess both traditional project delivery models like DBB, DB and

90

DBF models against the P3 model in attaining responsible sustainability for the project. It will also screen for a satisfactory level to opt for a P3 model in align with environmental, social, and governance goals of social infrastructure project. For every ESG criterion, a response indicator, the description, and measured question is provided as seen on appendix II. Once the social infrastructure project is considered suitable for a P3 model to achieve responsible sustainability, the most suitable type of P3 model is determined during the second stage of the procurement process.

In the second stage of the procurement process, this thesis developed a multi-criteria analysis tool, an AHP-PPP model, to aid in selecting the best P3 alternative model by integrating both quantitative and qualitative criteria in the evaluation. Only primary, mutually independent ESG criteria and VfM were considered for assessment to prevent potential inconsistencies arising from interdependent sub-criteria in the P3-ESG screening matrix. The effectiveness of the AHP-PPP model in practical applications has been demonstrated through real-life case studies where a comparative AHP-scale study was conducted using two AHP scales for pairwise comparison. The results remained consistent for both the fundamental AHP scale and the balanced-n scale, regardless of the presence of unequal weight distribution when comparing options with close preferences for more than two criteria, as suggested by Goepel [135].

The results of this proposed AHP-PPP model will depend on expertise and skills of the decision-making team and the specific ESG goals of each social infrastructure project. These factors will influence the preferences and decisions generated by the AHP-PPP model. Similarly, when applying the AHP-PPP model to projects, there is a lack of historical data to inform decisions for the progressive P3 model. This absence of past performance information poses a challenge in accurately assessing the significance of each ESG criterion. As a result, preference scoring from the input value from 1 to 9 relied more heavily on predictions and informed assumptions.

91

In conclusion, this thesis introduces a new approach to procuring sustainable P3 models by integrating non-financial ESG factors into P3 project screening and P3 model selection. Achieving a more environmentally and socially conscious Canada relies on successfully delivering sustainable social infrastructure projects, as they constitute the largest type of infrastructure projects delivered in Canada using P3 models. The thesis emphasizes evaluating P3 models as tools for responsible sustainability from the planning stage of P3 project procurement. It addresses the balanced need to think about social and environmental factors alongside financial considerations when evaluating P3 models. The achieved research objectives align with current demands for sustainable infrastructure development, contributing to societal benefits, environmental well-being, and a more responsible future for Canada's social infrastructure projects.

## 5.2. Recommendations

The following is a list of future research directions:

- 1. Explore incorporating potential risks into the AHP-PPP model, including understanding how to assign importance to the identified ESG risks.
- 2. Apply the P3-ESG screening matrix for indigenous social infrastructure P3 projects on reserve lands to identify their context-specific ESG considerations.
- 3. Conduct case studies on sustainable and responsible performance of progressive P3 models currently procured for transit and hospital projects to assist decision makers in making informed relative comparison of the progressive P3 models to other P3 models when using the AHP-PPP model.

These recommendations would contribute further to developing a relevant industry standard ESG-PPP decision-making framework, acknowledging the evolving dynamics of P3 projects and ESG considerations in different context, and promoting sustainable infrastructure development in Canada.

# Appendix I

# Sub-Criteria Selection (based on literature review)

The sub-criteria used for the screening matrix were selected according to the highest number of occurrences in the literature reviewed. To ensure the matrix is not exhausted with a lot of redundant sub-criteria only two were selected for each dimension. Therefore, the assessment focused on evaluating "green-related risks" and "green-adapted processes" within the broader process of forming green expertise partnerships.

Criteria	Proposed extrapolation of sub- criteria in relevance to project management and identified objectives and challenges in social infrastructure	Frequency of occurrence	No. of Occurrences
Environmental	Green expertise partnership	[114] [91] [52] [13] [30] [39] [147] [63] [112]	9
	Green funding	[148] [115] [50] [48] [116] [149]	6
	Green (climate) risk	[150] [151] [152]	3
	Circular economy: Green adapted processes	[116] [153]	2
Social	Community benefits agreements	[1] [63] [43] [94] [51]	5
	Consultation with indigenous peoples	[111] [152]	2
Governance	Transparency Accountability	[16] [49] [78]	3



# Appendix II

# Screening Matrix form

Criteria 1: Environmental: Green adoption incentives										
How does the P3 model incentivize and adopt environmentally friendly practices, technologies, and standards?										
What is	Green expertise partnership: The level of green expertise and Weighting									
being measured?	sustainability knowledge possessed by the private sector partner. This includes evaluating their track record in implementing environmentally friendly practices and their capacity to innovate in sustainable technologies and solutions.									
	Question asked?	Are there any environmenta	partnershi Il organizati	ps or collabora ons or experts ject sustainabi	tions with to leverage lity?	e expertise				
			2	3	4	5				
	Indicator:	No green expertise	Limited green expertise	Moderate green expertise	Substantia I green expertise	High green expertise				
What is	Green fund	ling: The flexib	ility in finan	cial partnershi	p, allowing	for the	Weighting			
being measured?	incorporation sustainabilit	on of green fina ty-linked loans,	ncing mech or perform	anisms such a ance-based co	as green bo ontracts.	nds,	16.7%			
	Question asked?Does the P3 model have environmental conscious funding options, like green bonds or sustainability-linked loans?									
		1	2	3	4	5				
	Indicator:	No Green Funding	Limited Green Funding	Partial Green Funding	Moderate Green Funding	High Green Funding				

## Criteria 2: Social: Community benefits: Community Impact Phases and Community engagement

How does the P3 model influence the nature and scope of community benefits agreements, emphasizing and prioritizing immediate gains or long-term sustainability and if they are developed collaboratively with community stakeholders including local SMEs?

What is	Long-Term CBA: Duration of community benefits for non-core services,									
being measured?	focusing on long-term social benefit engagement.									
	Question asked?Does the PPP model involve long-term integration of community benefits?									
		1	2	3	4	5				
	Indicator:	No Integration	Limited Integration	Partial Integration	Moderate Integration	High Integration				
What is	Collaborat	ceholders in	Weighting							
being	developing	leveloping CBAs.								
measured?	Question How does the P3 model ensure the CBA have feedback from									
	asked?	ked? the community?								
		1	2	3	4	5				
	Indicator:	No Community feedback	Limited feedback	Partial feedback	Moderate feedback	High feedback				

## Criteria 3: Governance: Stakeholder engagement: Transparency and Accountability

How does the P3 model influence the overall structure of decision-making processes within the project team and other stakeholders, involving the formulation, discussion, communication, finalization, and implementation of project agreements?

What is	Transparer	ncy: Level of tr	ansparency	/ in P3 model t	hrough ear	ly	Weighting			
being measured?	engagemer long-term s	ment, communication channels, decision-making structure, and m stakeholder management								
	Question asked?	How does the different phase	How does the P3 models ensure stakeholder engagement in different phases of the project?							
		1								
	Indicator:	No Transparency	Limited Transpare ncy	Partial Transparency	Moderate Transpare ncy	High Transparency throughout project life cycle				
What is	Accountab	ility: The degr	ee of consis	stency in meeti	ng project i	milestones	Weighting			
being measured?	and financia set goals ar	al targets. How nd objectives.	will project	team take res	oonsibility f	or achieving	16.7%			
	Question asked?	QuestionHow consistently does the P3 model meet projectasked?environmental, social objectives and financial targets, demonstrating a high level of accountability?								
		1 2 3 4 5								
	Indicator:         No         Limited         Partial         Moderate         High           Accountability         Accountab         Accountability         Account					High Accountability				

## Appendix III

## AHP-PPP Excel tool

## South Niagara Hospital – Pairwise Comparisons

To choose the preference judgment for comparing the criteria at level 2 of the hierarchy structure, the Excel tool provides a dropdown list of input scales

A. SAATY SCALE: Drop down list of values: 1,2,3,4,5,6,7,8,9

Level 2:



For this case, the environmental is considered to be two times important compared to the social criterion this indicates it is moderately important compared to the social, governance and value for money criterion.

B. BALANCED- N SCALE: The drop-down list of values includes: 1,1.36,1.8, 2.33, 3, 3.86, 5, 6.6, 9. This scale input, as indicated by the formula in Table 17 above, accounts for the number of criteria being compared, which in this case is 4 at level 2



Likewise, using the balanced-n scale, the environmental is considered to be two 1.36 times important compared to the social criterion this indicates it is moderately important compared to the social, governance and value for money criterion.

A. SAATY SCALE: Level 3: Alternative pairwise comparison



Governance	DBFM	DBFMO	PR.DBFM	PR.DBFMO	PPP bundle model		Normalized					PV	Criteria weights	
DBFM	1	0.5	0.5	0.5	1	1	0.125	0.076923	0.1	0.142857	0.2	1	0.129	1
DBFMO	2	1	0.5	0.5	1	]	0.25	0.153846	0.1	0.142857	0.2	1	0.169	×
PR.DBFM	2	2	1	0.5	1	]	0.25	0.307692	0.2	0.142857	0.2	1	0.220	
PR.DBFMO	2	2	2	1	1	]	0.25	0.307692	0.4	0.285714	0.2	1	0.289	
PPP bundle model	1	1	1	1	1	1	0.125	0.153846	0.2	0.285714	0.2	1	0.193	
Sum	8	6.5	5	3.5	5	¥								
Value for money	DREM	DBEMO			PPP bundle model	1	1		Normalize		0.046397	]	P\/	Critoria woights
DREM	1	1	0.5	0.5	1	-	0 140957	0 140057	0.1	0 140957	0.2	-	0.146	cinterna wergints
DBFMO	1	1	0.5	0.5	1	-	0.142857	0.142857	0.1	0.142857	0.2	-	0.146	- 1
PR.DBFM	2	2	1	0.5	1	1	0.285714	0.285714	0.2	0.142857	0.2		0.223	
PR.DBFMO	2	2	2	1	1	1	0.285714	0.285714	0.4	0.285714	0.2		0.291	
PPP bundle model	1	1	1	1	1		0.142857	0.142857	0.2	0.285714	0.2		0.194	
Sum	7	7	5	3.5	5	×				Cor	nsistenecy			_
											0.022526	-		

B. BALANCED – N SCALE: Level 3: Alternative pairwise comparison: The drop-down list of values includes: 1, 1.42, 1.91, 2.3, 3.22, 4.125, 5.29, 6.83, and 9. This scale input, as indicated by the formula in Table 17 above, accounts for the number of criteria being compared, which in this case is 5 at level 3





### B. BALANCED-N SCALE: Level 2




### Saskatchewan Joint Use School Projects – Pairwise comparison.

#### A. SAATY SCALE: Level 2

	Enviromental	Social	Governance	Value for Money	
Enviromental	1	0.5	0.333333333	1	
Social	2	1	0.5	1	
Governance	3	2	1	1	
Value for money	1	1	1	1	1
Sum	7	4.5	2.833333333	4	*



### B. BALANCED-N SCALE: Level 2

	Enviromental	Social	Governance	Value for money		Normalized			PV	Criteria weights		
Enviromental	1	0.735294118	0.55555556	1	1	0.193798	0.162127	0.178571	0.25	1	0.193	
Social	1.36	1	0.55555556	1	]	0.263566	0.220493	0.178571	0.25		0.227	
Governance	1.8	1.8	1	1	]	0.348837	0.396887	0.321429	0.25		0.335	]
Value for money	1	1	1	1	1	0.193798	0.220493	0.321429	0.25		0.246	]
Sum	5.16	4.535294118	3.11111111	4	<b>*</b>					· /		
					-				Co	nsistency 🖌		
						4.047		0.016		0.018		

### A. SAATY SCALE: Level 3



B. BALANCED-N SCALE: Level 3

Enviromental	DBFM	DBFMO	PR.DBFM	PR.DBFMO	PPP bundle model	Normalized PV Cr	riteria weights
DBFM	1	1	0.523560209	0.704225352	0.523560209	0.138122 0.171233 0.112647 0.137431 0.143984 0.141	/
DBFMO	1	1	0.704225352	1	0.704225352	0.138122 0.171233 0.151518 0.195151 0.193669 0.170	
PR.DBFM	1.91	1.42	1	1	0.704225352	0.263812 0.243151 0.215156 0.195151 0.193669 0.222	
PR.DBFMO	1.42	1	1	1	0.704225352	0.196133 0.171233 0.215156 0.195151 0.193669 0.194	
PPP bundle model	1.91	1.42	1.42	1.42	1	0.263812 0.243151 0.305522 0.277115 0.27501 0.273	
Sum	7.24	5.84	4.647785562	5.124225352	3.636236266	×,	
						5.032 0.008 0.007	
Social	DBFM	DBFMO	PR.DBFM	PR.DBFMO	PPP bundle model	Normalized PV C	riteria weights
DBFM	1	1	0.704225352	1	0.523560209	0.157978 0.157978 0.145849 0.184502 0.151512 0.160	1
DBFMO	1	1	0.704225352	1	0.523560209	0.157978 0.157978 0.145849 0.184502 0.151512 0.160	-
PR.DBFM	1.42	1.42	1	1	0.704225352	0.224329 0.224329 0.207106 0.184502 0.203794 0.209	
PR.DBFMO	1	1	1	1	0.704225352	0.157978 0.157978 0.207106 0.184502 0.203794 0.182	
PPP bundle model	1.91	1.91	1.42	1.42	1	0.301738 0.301738 0.29409 0.261993 0.289388 0.290	
Sum	6.33	6.33	4.828450704	5.42	3.455571123	¢ /	
						5.018 0.004 0.004	
Governance	DBFM	DBFMO	PR.DBFM	PR.DBFMO	PPP bundle model	Normalized PV Cr	riteria weights
DBFM	1	0.704225352	0.704225352	0.704225352	0.434782609	0.132275 0.125436 0.137431 0.137431 0.129138 0.132	/
DBFMO	1.42	1	1	1	0.523560209	0.187831 0.178119 0.195151 0.195151 0.155507 0.182	
PR.DBFM	1.42	1	1	1	0.704225352	0.187831 0.178119 0.195151 0.195151 0.209168 0.193	
PR.DBFMO	1.42	1	1	1	0.704225352	0.187831 0.178119 0.195151 0.195151 0.209168 0.193	
PPP bundle model	2.3	1.91	1.42	1.42	1	0.304233      0.340207      0.277115      0.297019      0.299	
Sum	7.56	5.614225352	5.124225352	5.124225352	3.366793522	Consistency 5.010 0.003 0.002	
Value for Money	DBFM	DBFMO	PR.DBFM	PR.DBFMO	PPP bundle model	Normalized PV Cr	riteria weights
DBFM	1	0.704225352	0.704225352	0.704225352	0.310559006	0.117925 0.116705 0.116705 0.155366 0.101427 0.122	/
DBFMO	1.42	1	1	0.704225352	0.523560209	0.167453 0.165721 0.165721 0.155366 0.170992 0.165	
PR.DBFM	1.42	1	1	0.704225352	0.523560209	0.167453 0.165721 0.165721 0.155366 0.170992 0.165	
PR.DBFMO	1.42	1.42	1.42	1	0.704225352	0.167453 0.235324 0.235324 0.22062 0.229996 0.218	
PPP bundle model	3.22	1.91	1.91	1.42	1	0.379717 0.316528 0.316528 0.313281 0.326594 0.331	
Sum	8.48	6.034225352	6.034225352	4.532676056	3.061904777	Consistence 5.022 0.006 0.005	

## Appendix IV

### Sensitivity Analysis Graphs

# South Niagara Hospital – Sensitivity Analysis

























## References

- [1] Government Canada, "Investing in Canada Building a better Canada," 15 December 2023.
  [Online]. Available: https://www.infrastructure.gc.ca/plan/about-invest-aproposeng.html. [Accessed 2 June 2022].
- [2] Government of Canada , "Investing in Canada Infrastructure Program," 20 December 2023. [Online]. Available: https://www.infrastructure.gc.ca/plan/icp-pic-INFC-eng.html. [Accessed 2 June 2022].
- [3] Infrastructure Canada, "Building pathways to 2050: Moving forward on the National Infrastructure Assessment," 2021. [Online]. Available: chromeextension://efaidnbmnnnibpcajpcglclefindmkaj/https://www.infrastructure.gc.ca/altformat/pdf/nia-eni/nia-eni-2-en1.pdf. [Accessed 03 06 2022].
- [4] Government of Canada, "Towards Canada's 2030 Agenda National Strategy," 15 June 2019. [Online]. Available: https://www.canada.ca/en/employment-socialdevelopment/programs/agenda-2030/national-strategy.html. [Accessed 7 June 2021].
- [5] G. H. Brundtland, "Report of the World Commission on Environment and Development: Our Common Future," 20 March 1987. [Online]. Available: chromeextension://efaidnbmnnnibpcajpcglclefindmkaj/https://sustainabledevelopment.un.org/ content/documents/5987our-common-future.pdf. [Accessed 8 June 2022].
- [6] Cassa Depositi e Prestiti, "Social Infrastructure: Sectoral Strategic Guideline," [Online]. Available: chromeextension://efaidnbmnnnibpcajpcglclefindmkaj/https://www.cdp.it/resources/cms/docu ments/CDP\_SOCIAL\_INFRASTRUCTURE\_ENG.pdf. [Accessed 1 July 2022].
- [7] A. W. Connors, An exploratory study of the use of Social Infrastructure Public-Private Partnership (P3s) for Public University Facilities, Florida: University of Central Florida, 2021.
- [8] Government of Canada, "Investing in Canada Plan funding stream: Social infrastructure," 2024. [Online]. Available: https://www.infrastructure.gc.ca/plan/si-is-eng.html. [Accessed 16 April 2024].
- [9] SP3CTRUM, "SP3CTRUM," 2024. [Online]. Available: http://www.p3spectrum.ca/project/. [Accessed 2 May 2022].
- [10] Canadian Council of Public Private Partnerships , "Types of P3s," [Online]. Available: https://www.pppcouncil.ca/why-p3s/types-of-p3s. [Accessed 1 October 2023].
- [11] Canadian Council for Public-Private Partnerships, "The Evolution of Canada's Model: P3 Policy Discussion paper," October 2023. [Online]. Available: chromeextension://efaidnbmnnnibpcajpcglclefindmkaj/https://www.pppcouncil.ca/getmedia/4 2b4a5fb-84df-4d76-8494-129dc1baa028/P3-Policy-Discussion-Paper\_October2023.pdf. [Accessed 2023].
- [12] D. Baxter, "PPP Evaluation Criteria: Procurement's Achilles Heel?," 8 September 2022. [Online]. Available: https://www.linkedin.com/pulse/ppp-evaluation-criteriaprocurements-achilles-heel-david-baxter/. [Accessed 15 December 2023].
- [13] S. Hussain and J. McKellar, "Exploring the Success of Social Infrastructure Public Private Partnerships: The Complex Case of Bridgepoint Active Healthcare in Ontario, Canada," *Public Works Management & Policy*, vol. 25, no. 3, pp. 259-280, 2020.

- [14] K. A. A. S. A. H. Mounir El Khatib, "Sustainable Project Management: Trends and Alignment," *Theoretical Economics Letters*, vol. 10, no. 6, p. 16, 2020.
- [15] Buy social canada, 2018. [Online]. Available: https://www.buysocialcanada.com/wpcontent/uploads/BSC\_socialprocurement\_R5\_EN\_SCREEN.pdf.. [Accessed 6 August 2023].
- [16] CBC News, 2 men charged with fraud after probe into \$300M St. Michael's Hospital redevelopment, Toronto: CBC, 2023.
- [17] The Canadian Council For Public-Private Partnerships, "Building North America's first fully digital Hospital," The Canadian Council For Public-Private Partnerships, Ontario, 2015.
- [18] The Council of canadians, "Public-private partnerships have no place in canada's postcovid just recovery," 12 November 2020. [Online]. Available: https://canadians.org/analysis/public-private-partnerships-have-no-place-canadaspost-covid-just-recovery/. [Accessed 2023 December 15].
- [19] Prevalent, "Navigating ESG and Third-Party Risk Management," Prevalent, 2023.
- [20] O. L. B. A. O. T. N. O. J. L. Ali Hosseini, "Selection Criteria for Delivery Methods for Infrastructure Projects," *Procedia - Social and Behavioral Sciences*, vol. 226, pp. 260-268, 2016.
- [21] Canadian Housing and Renewal Association, "What is Community Housing?," [Online]. Available: https://chra-achru.ca/what-is-communityhousing/#:~:text=Community%20housing%3A%20Also%20known%20as,part%20by%2 0provincial%20rent%20supplements.. [Accessed 8 February 2024].
- [22] Canada Mortage and Housing Corporation, "Canada's Housing Supply Shortage: Restoring affordability by 2030," 23 June 2022. [Online]. Available: https://www.cmhcschl.gc.ca/blog/2022/canadas-housing-supply-shortage-restoring-affordability-2030. [Accessed 2023 November 20].
- [23] Government of Alberta, "Fiscal plan: A Responsible Plan for a Growing Province," Government of Alberta, Alberta, 2024-2027.
- [24] Government of Ontraio, "More Homes Built faster," 25 October 2022. [Online]. Available: https://www.ontario.ca/page/more-homes-builtfaster#:~:text=Our%20government%20has%20committed%20to,next%20step%20to%2 0getting%20there.. [Accessed 8 February 2023].
- [25] Infrastructure Ontario, "Transit-Oriented Communities (Subways)," 2021. [Online]. Available: https://www.infrastructureontario.ca/en/what-we-do/development/transit-orientedcommunities-subways/. [Accessed 20 December 2023].
- [26] Government of British Columbia: Stronger BC for everyone, "Homes for People: An action plan to meet the challenges of today and deliver more homes for people, faster," 2023. [Online]. Available: https://news.gov.bc.ca/files/Homes\_For\_People.pdf.. [Accessed 20 January 2024].
- [27] Saskatchewan Housing Corporation, "The Saskatchewan Housing Action Plan 2022-2023," Saskatchewan Housing Corporation, Saskatchewan, 2022-2023.
- [28] J. Chava and P. Newman, "Stakeholder Deliberation on Developing Affordable Housing Strategies: Towards Inclusive and Sustainable Transit-Oriented Developments," Sustainability, vol. 8, no. 10, p. 1024, 2016.

- [29] J. Grant, "10 reasons affordable housing is hard to deliver," 31 August 2023. [Online]. Available: https://www.halifaxexaminer.ca/commentary/10-reasons-affordable-housingis-hard-to-deliver/. [Accessed 11 December 2023].
- [30] Ministere de l'Education, "The Educational Project Guide," Ministere de l'Education, Quebec, 2022.
- [31] Alberta , "Planning and building schools:Learn how Alberta school projects are planned and find projects approved for construction," [Online]. Available: https://www.alberta.ca/planning-and-building-schools. [Accessed 5 January 2024 ].
- [32] Government of Alberta: Alberta Education , "School Capital Manual for the 2022/23 School Year," Government of Alberta , Alberta , 2022.
- [33] The Canadian Council For Public-Private Partnerships, "P3's: Bridging the first nations infrastructure gap," The Canadian Council For Public-Private Partnerships, 2016.
- [34] ecoschools, "Our impact," ecoschools, [Online]. Available: https://ecoschools.ca/about/ourimpact/. [Accessed 06 February 2024].
- [35] Sustainable Schools: Saving our energy for education , "2023 Top Energy Performing School Boards Report," Climate Challenge Network, Ontario, 2023.
- [36] Infrastructure Canada, "Environmental Impact Assessment," 3 February 2023. [Online]. Available: https://www.infrastructure.gc.ca/alt-format/pdf/assessment-enviro-evaleng.pdf. [Accessed 5 January 2024].
- [37] First Nations Infrastructure Institute, "First Nations Infrastructure Institute: Closing the Gap," 2024. [Online]. Available: https://fnii.ca/. [Accessed 2024].
- [38] J. French, "Alberta no longer using P3 approach as preferred way to build schools," CBC News, 26 December 2022. [Online]. Available: https://www.cbc.ca/news/canada/edmonton/alberta-no-longer-using-p3-approach-aspreferred-way-to-build-schools-1.6697233. [Accessed 30 November 2023].
- [39] Canadian Institute, "Accelerating Innovative Approaches to Healthcare Facility Design and Delivery: Healthcare Infrastructure, Eastern Canada," 2024. [Online]. Available: https://www.canadianinstitute.com/canadian-healthcare-infrastructure/. [Accessed 20 January 2024].
- [40] J. D. S. J. M. Matthew J. Eckelman, "Life cycle environmental emissions and health damages from the Canadian healthcare system: An economic-environmental-epidemiological analysis," *PLoS Med*, vol. 15, no. 7, 2018.
- [41] G. R. a. C. Samuelson, "Deferred Hospital Maintenance in Canada: There is more to 'a building' than building it," HealthCareCAN, Ontario, 2016.
- [42] T. Roberts, "A depleted treasury making it difficult to keep up with health maintenance, says John Haggie," CBC News, 24 01 2018. [Online]. Available: https://www.cbc.ca/news/canada/newfoundland-labrador/health-maintenancedeferred-1.4501361. [Accessed 4 December 2023].
- [43] B. Burmahl and J. Morgan, "Maintenance staff plays key design and construction role: Reallife examples prove the value of maintenance staff involvement in health care construction projects," American Hospital Association: Health facilities management Magazine, 31 March 2022. [Online]. Available: https://www.hfmmagazine.com/articles/4424-maintenance-staff-plays-key-design-andconstruction-role. [Accessed 6 December 2023].

- [44] S. T. Tamur and K. R. Erzaij, "The effectiveness of project delivery systems in the optimal implementation of green buildings," in *IOP Conference Series: Materials Science and Engineering*, Iraq, 2021.
- [45] "Investing in Canada Plan funding stream: Public transit infrastructure," Government of Canada, 26 9 2023. [Online]. Available: https://www.infrastructure.gc.ca/plan/pti-itceng.html. [Accessed 3 March 2023].
- [46] Federation of Canadian Municipalities, "Public Transit," [Online]. Available: https://fcm.ca/en/focus-areas/public-transit. [Accessed 9 November 2023].
- [47] Infrastructure Canada, "Permanent, Integrated, and Locally Responsive: New Foundations for Public Transit Funding in Canada," Infrastructure Canada, https://www.infrastructure.gc.ca/alt-format/pdf/transit-transport/pptf-report-rapportfptc-en.pdf, 2023.
- [48] Canadian Urban Transit Association (CUTA), "CUTA calls for urgently needed transit funding in 2024 federal budget," 15 February 2024. [Online]. Available: https://cutaactu.ca/cutacalls-for-urgently-needed-transit-funding-in-2024-federalbudget/#:~:text=Public%20transit%20systems%20collectively%20faced,ability%20to%2 0serve%20communities%20effectively.. [Accessed 2 March 2024].
- [49] A. M. Jones, "Canadians want transparency, region-specific solutions in government's plan to fund all public transit: report," CTV News, 25 July 2023. [Online]. Available: https://www.ctvnews.ca/canada/canadians-want-transparency-region-specificsolutions-in-government-s-plan-to-fund-all-public-transit-report-1.6494411. [Accessed 9 January 2024].
- [50] D. U. S. T. D. Carlos Dominguez Ordonez, "Green bonds in Public-Private Partnerships," International institute for Sustainable Development, Winnipeg, 2015.
- [51] Statistics Canada, "Canada's Core Public Infrastructure Survey: Culture, recreations and sports facilities, and public social and affordable housing, 2020," Statistics Canada, 2022.
- [52] Canadian Infrastructure Report Card (CIRC), "Monitoring the State of Canada's Core Public Infrastructure," Canadian Infrastructure Report Card, Ottawa, 2019.
- [53] P. B. K. L. Kevin Quigley, "Cultural and Recreational Infrastructure in Nova Scotia: Future Investments and Governance," MacEachen and Dalhouse University, Nova Scotia, 2019.
- [54] F. A. J. N. G. G. P. Prita Amalia, "A systematic review of the interrelation of administrative, Environmental, Social and Governance of Public-Private Partnerships (PPP) Spaceport Project in Biak, Papua, Indonesia," *Journal of Infrastructure, Policy and Development*, vol. 7, no. 1, p. 1709, 2023.
- [55] J. L. Y. H. Caichuan Wang, "Construction of Performance Evaluation Model for the Operation and Maintenance of Government Enterprise Cooperation Infrastructure Projects under the ESG Concept," Social Security and Administration Management, vol. 4, no. 7, pp. 11-18, 2023.
- [56] Government of Canada, "Chapter 9: Special procurements," 23 November 2023. [Online]. Available: https://canadabuys.canada.ca/en/how-procurement-works/policies-andguidelines/supply-manual/chapter-9#\_9-60. [Accessed 20 July 2023].
- [57] PPP Canada, "A process guide for public sponsors," PPP Canada, Ontario, 2017.
- [58] "Competition Issues in Public-Private Partnerships," Organisation for Economic Co-operation and Development (OECD), 16 June 2014. [Online]. Available:

https://www.oecd.org/competition/competitionissuesinpublic-privatepartnerships.htm. [Accessed 13 April 2024].

- [59] The Canada council for public private partnerships, "Types of P3s," 2022. [Online]. Available: https://www.pppcouncil.ca/why-p3s/types-of-p3s. [Accessed 9 June 2023].
- [60] King and Spalding, "Delivering Infrastructure Projects under a PPP Model: A guide to achieving success in Road, Rail, Airport, Port< Healthcare, Education, Housing and Accomodation PPP projects in the Middle East," King and Spalding, United Arab Emirates, 2017.
- [61] Infrastructure Ontario, "Infrastructure Ontario Projects: P3/ Major projects," 2024. [Online]. Available: https://www.infrastructureontario.ca/en/what-wedo/projectssearch/?cpage=1&facets=projecttype%3Ap3major. [Accessed 2024 January 8].
- [62] Infrastructure Alberta, "Public-private partnerships (P3s)," 2024. [Online]. Available: https://www.alberta.ca/public-private-partnerships#jumplinks-1. [Accessed 7 January 2024].
- [63] Canadian Council for Public private partnerships, "Better, Faster, Smarter: Building a school legacy for Saskatchewan," Canadian Council for Public private partnerships, 2015.
- [64] J. French, "Alberta no longer using P3 approach as preferred way to build schools," 26 December 2022. [Online]. Available: https://www.cbc.ca/news/canada/edmonton/alberta-no-longer-using-p3-approach-aspreferred-way-to-build-schools-1.6697233. [Accessed 23 February 2024].
- [65] A. Birrell, "Three times P3s screwed over Saskatchewan," 6 January 2020. [Online]. Available: https://saskdispatch.com/articles/view/public-private-partnerships-p3s-screwed-oversaskatchewan. [Accessed 24 February 2023].
- [66] Ministry of Transportation and Infrastructure, "Pattullo Bridge Replacement Project: Procurement Options Report," Partnerships British Columbia, British Columbia, 2018.
- [67] Canadian Council for Public-private Partnerships, "Why P3s?," 2024. [Online]. Available: https://www.pppcouncil.ca/why-p3s. [Accessed 2024].
- [68] PPP Canada, "P3 Business Case Development Guide," [Online]. Available: https://www.yumpu.com/en/document/read/38096532/p3-business-casedevelopment-guide-ppp-canada. [Accessed 10 July 2022].
- [69] PPP Canada, "Identifying P3 potential: A guide for federal departments and agencies," PPP Canada, 2012.
- [70] A. Maly, "HOT or NOT: Does the Canadian Public-Private Partnership Legal Framework Attract Foreign Investment?," The Canadian Bar Association, 12 October 2021. [Online]. Available: https://www.cba.org/Sections/Construction-Law/Resources/Resources/2021/ConstructionEssayWinner2021. [Accessed 5 June 2022].
- [71] D. Penalver, "Public-Private Partnerships 2.0: Value for People and Value for Future," Getting Infrastructure Finance Right, 2 July 2020. [Online]. Available: https://blogs.worldbank.org/en/ppps/public-private-partnerships-20-value-people-andvalue-

future#:~:text=Public%2DPrivate%20Partnerships%202.0%3A%20Value%20for%20Peop le%20and%20Value%20for%20Future,-

Domingo%20Pe%C3%B1alver%20July&text=Traditionally%2C%20d. [Accessed 7 July 2023].

- [72] M. T. Ibanez, "Rightly done PPPs can be the right tool for green and resilient infrastructure," 01 July 2021. [Online]. Available: https://blogs.worldbank.org/en/ppps/rightly-done-pppscan-be-right-tool-green-and-resilient-infrastructure. [Accessed 06 July 2023].
- [73] Infrastructure Ontario, "Our Public-Private Partnerships (P3) Model," 2024. [Online]. Available: https://www.infrastructureontario.ca/en/what-we-do/major-projects/our-p3model/.
- [74] Alberta Infrastructure and Transportation , "Management Framework: Assessment Process," Alberta Infrastructure and Transportation , Alberta , 2006.
- [75] K. Reynolds, "Public-Private Partnerships in British Columbia," June 2018. [Online]. Available: chromeextension://efaidnbmnnnibpcajpcglclefindmkaj/https://columbiainstitute.eco/wpcontent/uploads/2019/11/Columbia-Institute-P3s-in-BC-2018-June-7-WEB.pdf. [Accessed 2022 February 2].
- [76] SaskBuilds, February 2016. [Online]. Available: chromeextension://efaidnbmnnnibpcajpcglclefindmkaj/http://www.saskbuilds.ca/alternativefinancing/SaskBuilds%20P3%20Project%20Assessment%20and%20Procurement%20G uideline%20-%20revised%202016.pdf. [Accessed 20 May 2022].
- [77] K. M. J. M. L. S. Ilan Dunsky, "Public-Private Partnerships: the case for enhanced collaboration in Canadian infrastructure projects," Lexpert| Business of Law, 29 September 2021. [Online]. Available: https://www.lexpert.ca/legal-insights/publicprivate-partnerships-the-case-for-enhanced-collaboration-in-canadian-infrastructureprojects/360008. [Accessed 2022 February 2].
- [78] M. Siemiatycki, "Public-Private Partnerships: Is a reassessment underway?," Ontario 360, Ontario , 2023.
- [79] Infrastructure Ontario, "Assessment of Innovation through AFP," Atlus Group, Ontario, 2015.
- [80] S. Enoch, "Is There Value in Value for Money Assessments: Testing the VfM Test in Western Canada," Canadian Centre for Policy Alternatives: Saskatchewan Office, Saskatchewan , 2021.
- [81] Infrastructure Ontario, "Trillium Health Partners Broader Redevelopment The Peter Gilgan Mississauga Hospital," [Online]. Available: https://www.infrastructureontario.ca/en/what-we-do/projectssearch/trillium-healthpartners-broader-redevelopment--the-peter-gilgan-mississauga-hospital/. [Accessed 25 January 2024].
- [82] Organisation for Economic Co-Operation and Development , "Getting Infrastructure Fight: The Ten Key Governance Challenges and Policy Options," Organisation for Economic Co-Operation and Development , Paris, 2016.
- [83] S. B. A.-K. M. S. M.-D. Ata ul Musawir, "Project governance and its role in enabling organizational strategy implementation: A systematic literature review," International Journal of Project Management, vol. 38, no. 1, pp. 1-16, 2020.
- [84] United nations economic commison for europe , "Guidebook on Promoting Good Governance in Public-Private Partnerships," United Nations, New York and Geneva, 2008.

- [85] S. H. Yingqi Song, "Influence of Project Governance Mechanisms on the Sustainable Development of Public-Private Partnership Projects: An Empirical Study from China," *Buildings*, vol. 13, no. 10, p. 2424, 2023.
- [86] Ontario Infrastructure and Lands Coporation, "Project Agreement: Finch west LRT project," Infrastructure Ontario, Ontario, 2018.
- [87] Alberta Infrastructure, "Schedule 9: Form of Tri-Party Agreeemnt," Alberta Infrastructure, Edmonton, 2010.
- [88] M. J. G. Geert Dewulf, "Responsive governance in PPP projects to manage uncertainty," *Construction Management and Economics,* vol. 38, no. 4, pp. 383-397, 2020.
- [89] S. S. Alie, "Project governance: #1 critical success factor," in *PMI® Global Congress 2015— North America*, Orlando, FL, 2015.
- [90] Canadian Construction Documents Committee, "Documents," [Online]. Available: https://www.ccdc.org/documents/. [Accessed 5 February 2024].
- [91] European PPP Extertise Centre, "Role and Use of Advisers in preparing and implementing PPP projects," European PPP Extertise Centre, Luxembourg, 2014.
- [92] B. M. C. M. H. G. Q. S. Xue Lin, "Stakeholders' influence strategies on social responsibility implementation in construction projects," *Journal of Cleaner Production*, vol. 235, pp. 348-358, 2019.
- [93] A. W.-F. a. J. Węgrzyn, "Understanding of Public–Private Partnership Stakeholders as a Condition of Sustainable Development," *Sustainability*, vol. 11, no. 4, p. 1194, 2019.
- [94] Constructive Edge and Canadian Council for Public private partnership, "P3s and Community Benefits Agreements in Canada," Constructive Edge, 2021.
- [95] ReNewCanada, 13 April 2021. [Online]. Available: https://www.renewcanada.net/feature/cbas-beneficial-building-trend/. [Accessed 15 January 2023].
- [96] Canada Green Building Council, "LEED: The global standard in green building," [Online]. Available: https://www.cagbc.org/our-work/certification/leed/. [Accessed 30 January 2023].
- [97] Canadian Green Building Council, "Zero Carbon Building Standards," 2024. [Online]. Available: https://www.cagbc.org/our-work/certification/zero-carbon-building-standard/. [Accessed 2024].
- [98] Canada Green Building Council, "Project Database," [Online]. Available: https://leed.cagbc.org/LEED/projectprofile\_EN.aspx. [Accessed 25 January 2023].
- [99] Canadian Council of Public private partnerships, "Celebrating the Best in Canadian Infrastructure: National Awards," 2023. [Online]. Available: https://www.pppcouncil.ca/what-do-we-do/national-awards. [Accessed 19 December 2023].
- [100] K. Friedrich, "A systematic literature review concerning the different interpretations of the role of sustainability in project management," *Management Review Quarterly*, vol. 73, no. 1, pp. 31-60, 2021.
- [101] R. C. Oliver Laasch, Principles of Responsible Management: Glocal Sustainability, Responsibility, Ethics, Cengage, 2015.
- [102] R. Alvarez Tinoco, C. Sato and R. Hasan, "Responsible project management: Beyond the triple constraints," *Modern Project Management*, vol. 4, no. 1, pp. 81-93, 2016.

- [103] C. B. a. B. M. Ira Kay, "Harvard Law School Forum on Corporate Governance: The Stakeholder Model and ESG," 14 September 2020. [Online]. Available: https://corpgov.law.harvard.edu/2020/09/14/the-stakeholder-model-and-esg/. [Accessed 27 February 2024].
- [104] J. X. O.-K. R. A. S. K. Isaac Akomea-Frimpong, "Public-private partnerships for sustainable infrastructure development in Ghana: A systematic review and recommendations," *Smart and Sustainable Built Environment*, vol. 12, no. 2, pp. 237-257, 2023.
- [105] S. H. X. L. Guoli Feng, "Project Sustainability and Public-Private Partnership: The Role of Government Relation Orientation and Project Governance," Sustainability, vol. 14, no. 2, p. 4724, 2022.
- [106] B. L. a. T. S. Bin Xue, "What Matters in Achieving Infrastructure Sustainability through Project Management Practices: A Preliminary Study of Critical Factors," *Sustainability*, vol. 10, no. 12, 2018.
- [107] K. V. T. B. Marlies Hueskes, "Governing public–private partnerships for sustainability: An analysis of procurement and governance practices of PPP infrastructure projects," *International Journal of Project Management*, vol. 35, no. 6, pp. 1184-1195, 2017.
- [108] J. L. D. W. H. L. G. W. J. Y. Bingsheng Liu, "Public–private partnerships: a collaborative framework for ensuring project sustainable operations," *Engineering, Construction and Architectural Management*, vol. 31, no. 1, pp. 1-26, 2024.
- [109] Organization for Economic Co-operation and Development , "ESG Investing: Practices, Progress and Challenges," Organization for Economic Co-operation and Development , Winnipeg, 2020.
- [110] K. W. J. C. J. D. a. L. H. Will Sloan, "Valuing Sustainability in Infrastructure Investments: Market Status, Barriers and Oppourtunities: A Landscape Analysis," WWF Switzerland and Cadmus Group, Switzerland, 2019.
- [111] Statistics Canada: Industrial Organization and Finance Division, "Environmental, Social and Governance Project and Indegenous Peoples: Engagement Report," Statistics Canada, Ottawa, 2023.
- [112] Government of Canada, "Basics of Environmental Assessment under the Canadian Environmental Assessment Act, 2012," 19 12 2023. [Online]. Available: https://www.canada.ca/en/impact-assessment-agency/services/policyguidance/basics-environmental-assessment.html. [Accessed 2 April 2022].
- [113] Association of Consulting Engineering Companies and Strategies 4 Impact, "Understanding Public private partnerships in Canada," Strategies 4 Impact, Ontario, 2006.
- [114] Canadian Council of Public Private Partnerships, "Building North America's first fully digital hospital," Canadian Council of Public Private Partnerships, Ontario .
- [115] Green Municipal Fund, "Helping municipalities create a sustainable and prosperous future," 2024. [Online]. Available: https://greenmunicipalfund.ca/.
- [116] Government of Canada , "Government of Canada: Green Bond Allocation : Report 2021-22," Government of Canada, Ottawa, 2023.
- [117] A. J. G. G. P. Prita Amalia Faiz, "A systematic review of the interrelation of administrative, Environmental, Social and Governance of Public-Private Partnerships (PPP) Spaceport Project in Biak, Papua, Indonesia," *Journal of Infrastructure Policy and Development,* vol. 7, no. 1, p. 1709, 2023.

- [118] A. C. a. J. Farrier, "Public–Private Partnerships for Environmental, Social, and Governance Projects: How Private Funding for Infrastructure Can Produce Mutual Benefits for Companies and the Public," *Leveraging Sustainable Infrastructure for Resilient Communities*, 2022.
- [119] PPIAF: Enabling Infrastructure Investment, "A PPP Screening Tool: PPP Support of the Nile Equatorial Lakes Subsidiary Action Program," Nodalis Conseil, Washington, 2017.
- [120] G. Odu, "Weighting Methods for Multi-Criteria Decision Making Technique," *Journal of Applied Sciences and Environmental Management*, vol. 23, no. 8, pp. 1449-1457, August 2019.
- [121] C. Yin, "A Brief Summary of Subjective Weighting Methods in MCDM," 3 September 2020. [Online]. Available: https://www.linkedin.com/pulse/brief-summary-subjectiveweighting-methods-mcdm-chonghua-yin/. [Accessed 17 March 2024 ].
- [122] S. A. a. M. P. B. Büşra Ayan, "A Comprehensive Review of the Novel Weighting Methods for Multi-Criteria Decision-Making," *Information*, vol. 14, no. 5, p. 285, 2023.
- [123] Partnerships British Columbia, "Methodology for Quantitative Procurement Options Analysis Discusion Paper," Partnerships British Columbia, British Columbia, 2014.
- [124] M. M. Hamed Taherdoost, "Multi-Criteria Decision Making (MCDM) Methods and Concepts," *Encyclopedia*, vol. 3, no. 1, pp. 77-87, 2023.
- [125] M. T. · . M. S. · . F. J. Santos-Arteaga, "Analytical hierarchy process: revolution and evolution," Annals of Operations Research, vol. 326, no. 2023, pp. 879-907, 2021.
- [126] M. H. S. A. a. T. Z. Mohamed Assaf, "A Multi-Criteria Decision-Making Model for Selecting the Best Project Delivery Systems for Offsite Construction Projects," *Buildings*, vol. 13, no. 2, p. 571, 2023.
- [127] M. I. A. Khalil, "Selecting the Appropriate Project Delivery Method Using AHP," *International Journal of Project Management,* vol. 20, pp. 469-474, 2002.
- [128] T. L. Saaty, "Decision Making The Analytical Hierachy and Network processes (AHP/ANP)," *Systems science and systems engineering*, vol. 13, no. 1, pp. 1-35, 2004.
- [129] Canadian council for Public Private Partnerships, "Celebrating the Best in Canadian Infrastructure: National Awards," 2024. [Online]. Available: https://www.pppcouncil.ca/what-do-we-do/national-awards. [Accessed 2023 June 1].
- [130] S. R. A. P. a. B. D. Zorica Dodevska, "When Fairness Meets Consistency in AHP Pairwise Comparisons," *Mathematics*, vol. 11, no. 3, 2023.
- [131] R. I. S. A. H. I. I. K. Ummu Hani' Hair Zaki, "Prioritized Text Detergent:Comparing Two Judgment Scales of Analytic Hierarchy Process on Prioritizing Pre-Processing Techniques on Social Media Sentiment Analysis," *Baghdad Science Journal*, vol. 21, no. 2, pp. 662-683, 2024.
- [132] K. Goepel, "Implementation of an Online Software Tool for the Analytic Hierarchy Process (AHP-OS)," IJAHP Article: Goepel/ Implementation of an online software tool for the Analytic Hierarchy Process (AHP-OS)International Journal of the Analytic Hierarchy Process, vol. 10, no. 3, pp. 469-487, 2018.
- [133] K. D. Goepel, "Comparison of Judgment Scales of the Analytical Hierarchy Process A New Approach," International Journal of Information Technology & Decision Making, vol. 18, no. 2, pp. 445-463, 2019.
- [134] T. L. Saaty, "How to make a decision: The analytic hierarchy process," *European Journal of Operational Research*, vol. 48, no. 1, pp. 9-26, 1990.

- [135] K. D. Goepel, "BPMSG: Comparison of Judgment Scales of the Analytical Hierarchy Process -A New Approach," 20 May 2018. [Online]. Available: chromeextension://efaidnbmnnnibpcajpcglclefindmkaj/https://bpmsg.com/wordpress/wpcontent/uploads/2018/05/2018-05-10-AHP-Judgm-Scales-blog.pdf. [Accessed 17 December 2023].
- [136] M. S. F. J. S.-A. Madjid Tavana, "Analytical hierarchy process: revolution and evolution," *Annals of Operations Research,* vol. 326, no. 2023, pp. 879-907, 2021.
- [137] Niagara Health , "What is the South Niagara Hospital?," 2024. [Online]. Available: https://www.niagarahealth.on.ca/site/south-niagara-about. [Accessed 2024].
- [138] A. C. Rosas, "Construction of new hospital in Niagara begins after concerns over staffing, urgent care closures," 19 july 2023. [Online]. Available: https://www.cbc.ca/news/canada/hamilton/south-niagara-hospital-construction-1.6910134.
- [139] CCPPP, "Nine projects shortlisted for 2023 National P3 Awards," 2024. [Online]. Available: https://www.pppcouncil.ca/news/nine-projects-shortlisted-for-2023-national-p3awards. [Accessed 2024].
- [140] Plenary , "South Niagara Hospital," 2024. [Online]. Available: https://plenary.com/project/south-niagara-hospital. [Accessed 2024].
- [141] Infrastructure Ontario, "South Niagara Hospital," 2024. [Online]. Available: https://www.infrastructureontario.ca/en/what-we-do/projectssearch/south-niagarahospital/#pAboutTheProject. [Accessed 2024].
- [142] WELL, "WELL v2," 2023. [Online]. Available: https://v2.wellcertified.com/en/wellv2/overview. [Accessed 30 March 2024].
- [143] Metrolinx, "Finch West LRT," 2024. [Online]. Available: https://www.metrolinx.com/en/projects-and-programs/finch-west-lrt/what-werebuilding. [Accessed 2024].
- [144] City News Everywhere, "Metrolinx staff 'very confident' Finch West LRT will open in 2024 after recent work," 27 December 2023. [Online]. Available: https://toronto.citynews.ca/2023/12/27/finch-west-lrt-2024-opening-metrolinx/. [Accessed 18 Januray 2024].
- [145] Infrastructure Ontario, "Value for money Assessment: Finch west light rail transit," Infrastructure Ontario, Ontario, 2018.
- [146] Infrastructure Ontario, "Finch west LRT project agreement," Infrastructure Ontario, Ontario, 2018 .
- [147] Government of Canada, "Private Sector Engagement for Sustainable Development," 6 1 2023. [Online]. Available: https://www.international.gc.ca/worldmonde/issues\_development-enjeux\_developpement/prioritiespriorites/fiap\_private\_sector-paif\_secteur\_prive.aspx?lang=eng. [Accessed 1 12 2023].
- [148] PwC Canada , "Making the most of green tax and incentive programs," 2022. [Online]. Available: https://www.pwc.com/ca/en/today-s-issues/environmental-social-andgovernance/green-tax-and-incentive-programs.html. [Accessed 2023].
- [149] Government of Canada , "Government of Canada: Green Bond Framework," Government of Canada, Ottawa, 2022.

- [150] "The Green Book: Central government guidance on Appraisal and Evaluation," 2018. [Online]. Available: https://iaac-aeic.gc.ca/050/documents/p65505/125818E.pdf. [Accessed 2024].
- [151] Bank of Canada , "Bank of Canada Disclosure of Climate-Related Risks 2022," Bank of Canada , Ottawa, 2022.
- [152] Council of Canadian Academies, "Canada's Top Climate Change Risks:The Expert Panel on Climate Change Risks and Adaptation Potential," Council of Canadian Academies, Ottawa, 2019.
- [153] Delphi Group , "Circular Economy & The built environment sector in Canada," Delphi Group , Ottawa, 2021.